Horsley Drive Stage 2 -Development Management Agreement

Dated 30 July 2018.

Western Sydney Parklands Trust (ABN 85 202 544 800) ("Owner")

CIP/CH (Horsley Drive) Pty Limited (ACN 627 824 97)) ("Developer")

Commercial & Industrial Property Pty Ltd (ABN 88 105 081 865)

and

Charter Hall Holdings Pty Limited (ABN 15 051 363 547)

(together "Guarantor")

Horsley Drive Stage 2 - Development Management Agreement Contents

Details				
General terms				
1	Interpretation	8		
1.1 1.2 1.3 1.4	Definitions Construction Headings Severability	8 17 18 18		
2A	Conditions subsequent	18		
2A.1 2A.2	1 Condition subsequent			
2	Appointment and term			
2.1 2.2 2.3	Appointment Term Exclusivity	19 19 19		
3	The Developer's role and responsibilities	20		
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8	Marketing and management responsibilities Leasing responsibilities Reimbursement of costs Access Road Costs Drainage Basin Costs Biodiversity Corridor Mitigation Measures Urban Farming Irrigation Water Reuse Scheme Developer's warranties	20 21 21 21 22 22 22 23 24		
4	The Owner's role and responsibilities	24		
4.1 4.2 4.3	Not used The Owner's role and responsibilities Removal of expired covenants	24 24 25		
5	DMA Sunset Date	25		
5.1 5.2 5.3 5.4	Owner's right to terminate Extension of DMA Sunset Date Conditions precedent to extension of DMA Sunset Date Delay caused by the Owner	25 25 26 26		
6	Site Works	27		
6.1 6.2 6.3 6.4 6.5	Principal Contractor External Infrastructure Works Internal Site Works Design and construction of Site Works Practical completion of works	27 27 28 28 29		
6.6 6.7	Variations to the Site Works Defects rectification			

6.8	Estimated Internal Site Works costs	30 30
6.9	Internal Site Works Budget and Program Health Safety and Environmental matters	
6.10 6.11	Health Safety and Environmental matters Site Construction Licence	
7	Development	
7.1	Supply of the Development Land for development	
7.2	Staging	
8	Developments on the Development Land	
8.1	Submission of Development Proposals	
8.2	Development Proposal	33
8.3	Development Proposal Criteria	34
8.4	Owner to enter into the Agreement for Ground Lease, Ground Lease and Construction Licence	25
8.5	Compliance with Agreement for Ground Lease	35 37
8.6	Termination of Agreement for Ground Lease	38
8.7	Construction Licence	38
8.8	Measurement of the Development Site	39
8.9	Multi-party Side Deed	39
8.10	Ownership of fixtures	39
9	Approvals	39
9.1	The Developer responsible	39
9.2	Approval Process	40
9.3	Development of a Development Site	41
10	Construction	41
10.1	Complete Development	41
10.2	Comply with Requirements	41
10.3	Insurance	42
11	Payment	42
11.1	Upfront Development Payment	42
11.2	Take-up Fee	42
11.3	Upfront Land Payment	43
11.4 11.5	Leasing Commission Construction Licence Fee	45 45
11.6	Tax invoices	45 45
12A	NSW Code and Implementation Guidelines	45
12A.1	Primary Obligation	46
12A.2	Access and information	40
12A.3	Sanctions	47
12A.4	Compliance	48
12A.5	General	49
12	Development Masterplan	49
12.1	Development Masterplan	
13	Access	50
13.1	Owner must permit access to Undeveloped Land	50
13.2	Access to adjoining land	51
13.3	Battering	51
		2

13.4 13.5	Conditions Release and Indemnity		
13.6	Undeveloped Land	51 52	
14	Environmental monitoring	52	
14.1	Developer accepts Development Site	52	
14.2	Notice of Contamination	53	
14.3	Owner gives no warranties	53	
15	Project Control Group	53	
15.1	Creation of Project Control Group	53	
15.2	Meetings of Project Control Group	54	
15.3	Conduct of Project Control Group meetings	54	
15.4	Reporting	55 55	
15.5	Costs of Project Control Group		
16	Single Sites	55	
16.1	Single Site for each Development Site	55	
16.2	Development Site Contributions or Levies	56	
17	No assignment by the Developer	57	
17.1	Consent	57	
18	No sale by Owner	58	
19	Dispute resolution	59	
19.1	Notification	59	
19.2	The reply	59	
19.3	The dispute	60 60	
19.4			
19.5	Expert	60	
20	Default	60	
20.1	Termination Events	60	
20.2	Termination	61	
20.3	Obligations on termination	61	
20.4	No other right to terminate	61	
21	Notices	62	
21.1	General	62	
21.2	Method of service	62	
21.3	Address for service	62	
21.4	Service by post	63	
21.5	Service by facsimile	63 63	
21.6	Service by email Form received	63	
21.7 21.8	Service after hours	63	
22A	GST	63	
22	Representations and warranties	64	
		64	
22.1			
22.2	Developer and Guarantor representations and warranties	64	

23	Miscellaneous	65	
23.1	Legal costs	65	
23.2	Stamp Duty	65	
23.3	Amendment	65	
23.4	Waiver and exercise of rights	66	
23.5	Rights cumulative	66	
23.6	Further assurance	66	
23.7	Computation of time	66	
23.8	Governing law and jurisdiction	66	
23.9	Joint and several liability	66	
23.10	Counterparts	66	
23.11	Effect of execution	66	
23.12	Entire understanding	66	
23.13	No agency	67	
23.14	Owner must execute	67	
23.15	Exercise of power	67	
24	Guarantee	67	
24.1	Guarantee	67	
24.2	Indemnity	68	
24.3	Payment	68	
24.4	Performance	68	
24.5	Effect	68	
24.6	Primary obligations	68	
24.7	No limitation	68	
24.8	Guarantor liability continues	69	
24.9	Purpose	69	
24.10	Survival	69	
24.11	Guarantor may not transfer	69	
25	Confidentiality	69	
25.1	Confidential	69	
25.2	Exceptions	69	
25.3	Overriding Obligations and Rights	69	
26	No merger	70	
27	Interest	70	
Signin	g page	71	
Sched	ule 1 –Development Proposal Criteria	73	
Sched	ule 2 – Plan of Development Land, Site and Adjoining Land	76	
Sched	ule 3 – Development Masterplan	77	
Sched	ule 4 – Calculations	78	
Sched	ule 5 – Internal Site Works Design Brief	81	
	Schedule 6 – External Infrastructure Works Design Brief		
Schedule 7 – Leasing Commission			
	Schedule 8 - Multi-Party Side Deed Principles		
	Schedule 9 – Form of Ministerial Consent		
Schedule 9 – Form of Ministerial Consent 8 Annexure A – Pro Forma Agreement for Ground Lease 8			
Annexure B – Pro Forma Construction Licence 8			

Annexure C – Site Construction Licence

Horsley Drive Stage 2 - Development Management Agreement

Details

Parties	Owner, Developer,	Guarantor
Owner	Name	Western Sydney Parklands Trust
	ABN	85 202 544 800
	Address	Level 7, 10 Valentine Avenue, Parramatta NSW 2150
	Telephone	+61 2 9895 7500
	Fax	+61 2 9895 7580
	Attention	Kerry Jahangir
Developer	Name	CIP/CH (Horsley Drive) Pty Limited
	ACN	627 824 971
	Address	Suite 59 Jones Bay Wharf (Upper Deck) 26-32 Pirrama Road Pyrmont NSW 2009
		and
		Level 20, 1 Martin Place, Sydney NSW 2000
	Telephone	+61 2 9506 1400
	Fax	+61 2 9506 1499
	Attention	"Company Secretary for Commercial & Industrial Property Pty Ltd"
Guarantor	Name	Charter Hall Holdings Pty Limited
	ABN	15 051 363 547
	Address	Level 20, 1 Martin Place, Sydney NSW 2000
	Telephone	+61 2 8651 9000
	Fax	+61 2 8295 8659
	Attention	Company Secretary for Charter Hall Holdings Pty Limited

	Name		Commercial & Industrial Property Pty Ltd
	ABN		88 105 081 865
	Addres	s	Suite 59 Jones Bay Wharf (Upper Deck) 26-32 Pirrama Road Pyrmont NSW 2009
	Teleph	one	+61 2 9506 1400
	Fax		+61 2 9506 1499
	Attentio	on	Company Secretary
Recitals	A The Owner is the registered proprietor of the Site.		
	В	The Owner agrees to make the Site available for commercial development.	
	с		r agrees to appoint the Developer as the exclusive of the Site on the terms and conditions contained eement.
	D The Guarantor agrees to guarantee the performance of the Developer under this Agreement.		
Governing law	New South Wales		
Date of agreement	See Signing page		

General terms

1 Interpretation

1.1 Definitions

In this Agreement:

Access Road has the meaning given to that term in clause 3.4(a).

Access Road Works has the meaning given to that term in clause 12A.1(a).

Agreement means this development management agreement and includes any appendices, schedules or annexures.

Agreement for Ground Lease means an agreement entered into between the Owner, Nominated Developer and an Approved Tenant for a Development Site under clause 8.4.

Allowable Delay Event means each of the following:

- (a) any delay in the carrying out and completion of the Site Works caused or contributed to by:
 - (i) inclement weather or conditions resulting from inclement weather;
 - latent conditions (for the avoidance of doubt, latent conditions do not include any conditions to the extent identified in reports available to the Developer, or otherwise known to the Developer at the date of this Agreement);
 - (iii) any combination of workers or strikes or lockouts or shortage or lack of materials or equipment (where such lockouts or shortages have resulted from industrial action) affecting the Site Works;
 - (iv) by reason of any earthquake, act of God, natural disaster, fire, riots, civil commotion, malicious damage, sabotage, act of public enemy, war, revolution, radioactive contamination or damage by aircraft or articles dropped from aircraft; or
 - (v) by reason of any other matter, cause or thing beyond the reasonable control of the Developer;
- (b) any delay to the extent that such delay is caused or contributed to by the Owner (other than by the Owner acting in accordance with this Agreement);
- (c) any delay by the Relevant Authority acting beyond its relevant statutory time limits, including the Council in issuing relevant Approvals or registration of plans of subdivision, consolidation or amalgamation;
- (d) any delay as a result of any proceeding taken by neighbouring owners or occupiers of land adjoining the Development Land in respect of the Development Land to the extent that such proceedings have not been caused by any default by the Developer under this Agreement or failure

by the Developer to take reasonable steps to consult with neighbouring owners of occupiers.

Approval includes the Initial Development Consent, any further modification to the Initial Development Consent and any other permit, licence, consent, certificate or other approval or exemption obtained or required to be obtained from a Relevant Authority in relation to:

- (a) the Development of the Development Land;
- (b) the Building Works;
- (c) the Internal Site Works; or
- (d) the External Infrastructure Works.

Approved Development Proposal means a Development Proposal or Revised Proposal approved by the Owner under clause 8.3. All documentation provided by the Developer as a part of that Development Proposal or Revised Proposal form part of the Approved Development Proposal.

Approved Permitted Use means general industrial, light industrial, warehouse and distribution and ancillary office land uses as defined under the Planning Documentation, but excludes container parks and storage areas, other than those that are ancillary to the approved use.

Approved Tenant means a tenant under an Approved Development Proposal that:

- (a) is not Insolvent;
- (b) has not been convicted of a criminal offence which carries a maximum sentence of equal to or more than 12 months in prison or multiple sentences that add up to equal to or more than 12 months in prison;
- (c) has not been the subject of a formal adverse finding by the Independent Commission against Corruption for conduct which, if prosecuted, carries a maximum sentence of equal to or more than 12 months in prison;
- (d) (combined with any proposed guarantor), is of sufficient financial standing to perform its obligations under the Agreement for Ground Lease and the Ground Lease; and
- (e) is otherwise ready, willing and able to perform its obligations under the Ground Lease.

Assignment Tests means that a person:

- (a) is not Insolvent;
- (b) has not been convicted of a criminal offence which carries a maximum sentence of equal to or more than 12 months in prison or multiple sentences that add up to equal to or more than 12 months in prison;
- (c) has not been the subject of a formal adverse finding by the Independent Commission against Corruption for conduct which, if prosecuted, carries a maximum sentence of equal to or more than 12 months in prison;
- (d) (combined with any proposed guarantor) is of sufficient financial standing to perform its obligations under this Agreement;

- (combined with any proposed guarantor) has demonstrable experience in successfully delivering and marketing comparable developments to the Development;
- (f) has an approach to environmental issues in relation to the Development consistent with the requirements of this Agreement;
- (g) has a delivery methodology in terms of design, construction, finance, management and maintenance suitable to effect the Development; and
- (h) is otherwise ready, willing and able to carry out the Development and otherwise perform the obligations of the Developer under this Agreement.

Building Works means the design, development, construction and completion of buildings and works on the Development Land or a Development Site but excluding the Site Works, undertaken by the Nominated Developer on behalf of an Approved Tenant.

Business Day means a day on which trading banks are opened for general banking business in Sydney.

Commencement of Construction in respect of each Development means the date the Nominated Developer (or a relevant appointed subcontractor) commences physical works on the relevant Development Site.

Contamination means the presence in, on or under land, air or water of a substance (including a soil, liquid, gas, odour, heat, sound, vibration or radiation) at a concentration above the concentration at which the substance is normally present, on or under (respectively) land, air or water in the same locality, being a presence that presents a risk of harm to human health or any other aspect of the Environment and **Contaminant** has a corresponding meaning.

Construction Licence means the licence in respect of a Development Site entered into between the Owner (as licensor) and the Nominated Developer (as licensee) under clause 8.4.

Construction Licence Fee means the fee calculated as set out in Part 5 of Schedule 4.

Corporations Act means the Corporations Act 2001 (Cth).

Design Documentation means schematic drawings, detailed plans and specifications for a Development and/or the Site Works (as applicable).

Developable Site Area means that part of the Development Land that is not, at the relevant time, subject to any Construction Licence that has commenced or a Ground Lease.

Developer's Agents means the Developer's authorised representatives, consultants, employees, agents, contractors or invitees.

Developing Site Areas means that part of the Development Land that is, at the relevant time, subject to a Construction Licence of a Development Site that has commenced or a Ground Lease.

Development means the design, development, construction and completion of buildings and works on the Development Land or a Development Site, as the case may be, in accordance with the relevant Approved Development Proposal and Agreement for Ground Lease and includes the Building Works and the Internal Site Works but does not include the External Infrastructure Works.

Development Land means that part of the Site that is able to accommodate built form projects and excludes the Access Road, Drainage Basins, common areas, land that is required to be dedicated to authorities or any other part of the Site affected in a similar way, and as indicated in the plan attached in Schedule 2.

The parties acknowledge that the total site area of the Development Land:

- (a) is approximately 142,784 square metres as at the date of this Agreement but will be subject to survey which may be carried out before or after the making of this Agreement; and
- (b) may change as a result of further design or any change to the Development Masterplan, subject to the Owner's approval in accordance with this Agreement.

Development Masterplan means the master plan for the development of the Development Land and includes each "Updated Development Masterplan" as that term is defined in clause 12.1.

Development Proposal means a written submission or proposal from the Developer to the Owner for Development of a particular Development Site, which proposal must be prepared in accordance with clause 8.2.

Development Proposal Criteria means the criteria set out in Schedule 1 of this Agreement.

Development Site means an area of the Development Land:

- (a) as indicated on a site plan for a Development in an Approved Development Proposal; or
- (b) which has been or is being developed by the Nominated Developer pursuant to an Approved Development Proposal.

Development Rights has the meaning given in clause 2.3(b).

DMA Sunset Date means the date which is as extended under clause 5.2.

Drainage Basins means the drainage infrastructure servicing the Site being:

- (a) Northern Basin detention basin on Lot 17 DP 13961;
- (b) Western basin detention basin on Lot 7 as shown in the plan in Schedule 2;
- (c) Eastern basin on-site stormwater detention/bio retention basin on Lot 6 as shown in the plan in Schedule 2;
- (d) Eastern channel channel within Lots 4 and 5 as shown in the plan in Schedule 2;
- (e) Western channel channel between Lots 2 and 3 as shown in the plan in Schedule 2;
- (f) Southern channel channel south of Lots 1 and 2 as shown in the plan in Schedule 2.

Drainage Basin Works means that portion of the Internal Site Works that relates to the construction of the Drainage Basins.

Effective Date means the day after the date of the Developer's notice under clause 2A.2(c) accepting the Minister's consent.

Environment includes all aspects of the surroundings of human beings:-

- (a) ecosystems and their constitute parts, including people and communities, natural and physical resources;
- (b) the qualities and characteristics of locations, places and areas; and
- (c) the social, economic, aesthetic and cultural aspects of a thing mentioned in paragraphs (a) or (b).

Environmental Law includes any Law relating to any aspect of the Environment.

External Infrastructure Works means the design and construction of the infrastructure works specified in the External Infrastructure Works Design Brief required to service the Development Land having regard to the requirements, guidelines, scope and extent of such works specified in the External Infrastructure Works Design Brief including delivery of Services to the boundary of the Site but excluding the provision of temporary services for construction purposes, gas services and remediation works.

External Infrastructure Works Design Brief means the design brief set out in Schedule 6.

Form of Ministerial Consent means the Minister's consent in the form of consent attached in Schedule 9.

GFA means the sum of the floor area of each floor of a building measured from the internal face of external walls, or from the internal face of walls separating the building from any other building, measured at a height of 1.4 metres above the floor, and includes:

(a) the area of a mezzanine,

but excludes:

- (b) any area for common vertical circulation, such as lifts and stairs;
- (c) vehicular access, loading areas, garbage and services;
- (d) plant rooms, lift towers and other areas used exclusively for mechanical services or ducting;
- (e) car parking to meet any requirements of the consent authority (including access to that car parking);
- (f) any space used for the loading or unloading of goods (including access to it);
- (g) terraces and balconies with outer walls less than 1.4 metres high; and
- (h) voids above a floor at the level of a storey or storey above.

GIPA Act means the Government Information (Public Access) Act 2009 (NSW).

Ground Lease means the ground lease to be granted to the Approved Tenant in accordance with an Agreement for Ground Lease.

Ground Rent means the total rent payable by an Approved Tenant under a Ground Lease.

GST means GST within the meaning of the GST Act and includes penalties and interest.

GST Act means A New Tax System (Goods & Services Tax) Act 1999 (Cth).

Guarantor means Charter Hall Holdings Pty Limited (ABN 15 051 363 547) and Commercial & Industrial Property Pty Ltd (ABN 88 105 081 865) and includes their respective successors and permitted assigns.

Implementation Guidelines means the Implementation Guidelines to the NSW Code of Practice for Procurement: Building and Construction (July 2013) (as enforced pursuant to any Practice Direction issued pursuant to the Implementation Guidelines).

Initial Development Consent means the development consent issued by the Minister for Planning for the development of the Site known as "SSD 7664 Horsley Drive Business Park Stage 2".

Initial Ground Rent means the annual rent payable by an Approved Tenant under a Ground Lease for the first year of the term of the relevant Ground Lease, calculated in accordance with Part 2 of Schedule 4.

Insolvent means in relation to a body corporate:

- (a) a "controller" (as defined in the Corporations Act), Manager, Trustee, Receiver, Receiver and Manager, Administrator or similar officer is appointed in respect of the body corporate or any asset of the body corporate;
- (b) a liquidator or provisional liquidator is appointed in respect of the body corporate;
- (c) an application is made to the court for an order, a court makes an order, a meeting is convened or a resolution is passed, for the purpose of:
 - appointing a person referred to in paragraph (a) or (b);
 - (ii) winding up or deregistering the body corporate; or
 - (iii) proposing or implementing a scheme of arrangement (other than a scheme of arrangement relating to a reconstruction or amalgamation while solvent),

other than an application which is withdrawn, dismissed or set aside within 21 days after it is made; and

(d) the body corporate enters into, resolves to enter into or proposes a reorganisation, moratorium or other form of administration involving an arrangement, composition or compromise with, or assignment for the benefit of, its creditors generally or any class of them, other than for the purposes of a reconstruction or amalgamation while solvent.

Internal Site Works means the demolition, design, development and construction of the infrastructure works as specified in the Internal Site Works Design Brief and the Urban Farming Irrigation Water Reuse Scheme, including bulk earth works, compaction to accommodate standard high level pad footings,

remediation, internal access road, stormwater works and Drainage Basins, retiring of biodiversity offset credits, reticulated Services to each Development Site, retaining walls (if applicable), dedication of the internal access road, temporary services required for construction purposes and batters.

Internal Site Works Budget means

Internal Site Works Design Brief means the design brief set out in Schedule 5.

Law means any statute, ordinance, code, regulation, by-law, local law, official directive, order, instrument, undertaking, obligation or applicable judicial administrative or regulatory decree judgment or order and includes the conditions and standards authorisations, licences, permits, consents, assurances, bonds or similar requirements including all applicable standards and obligations under the common law.

Lease Value means the lease value of the Development Land or Development Site calculated in accordance with Part 1 of Schedule 4.

Lease Value Rate means the lease value rate $(\$/m^2)$ of the Development Land or Development Site calculated in accordance with Part 1 of Schedule 4.

Leasing Commission means a lease commission for the relevant Agreement for Ground Lease, calculated in accordance with Schedule 7.

LRS means the New South Wales Land Registry Services.

Minister means the Minister for the Environment. As at the date of this Agreement the Minister is the Honourable Gabrielle Upton MP.

Net Land Payment has the meaning given to that term in clause 11.3(d).

Nominated Developer means the "Developer" entity nominated in respect of an Agreement for Ground Lease under clause 8.2(a), which may be the Developer or a Nominee.

Nominee means each:

- (a) Related Body Corporate of the Developer; or
- (b) a company or trust which is a joint venture company or trust in which the Developer, Commercial & Industrial Property Pty Limited ABN 88 105 081 865, Charter Hall Holdings Pty Limited ABN 15 051 363 547 or their Related Body Corporate participates; or
- (c) company or trust in which the Developer, Commercial & Industrial Property Pty Limited ABN 88 105 081 865, Charter Hall Holdings Pty Limited ABN 15 051 363 547 or their Related Body Corporate has a beneficial interest.

Northern Basin means the detention basin proposed in part Lot 17 in DP13961.

NSW Code means the NSW Government's Code of Practice for the Building and Construction Industry.

Outgoings means the following expenses payable in relation to the Development Land or the relevant Development Site:

(a) costs of insurances of the Development Land or the relevant Development Site;

- (b) costs of maintaining the Development Land in the state or condition as at the date of this Agreement; and
- (c) any costs incurred by the Owner in compliance with any Laws or requirements in respect of the Development Land or a Development Site.

Owner's Agents means the Owner's authorised representatives, consultants, employees, agents, contractors or invitees.

Planning Documentation means the Initial Development Consent and all documentation prepared and provided to the Relevant Authority as part of the application for the Initial Development Consent.

Practical Completion in relation to a Development has the meaning given to "Practical Completion" under the relevant Agreement for Ground Lease.

Practical Completion in relation to the External Infrastructure Works is achieved when:

- the relevant works are complete except for minor omissions and defects which do not prevent the works from being reasonably capable of being used for their intended purpose without unreasonable interruption or interference; and
- (b) the relevant requirements set out in the External Infrastructure Works Design Brief are satisfied.

Pro-Forma Agreement for Ground Lease means the forms of agreements for ground lease and ground lease as set out in Annexure A.

Pro-Forma Construction Licence means the forms of licences as set out in Annexure B.

Project Control Group means the group established under clause 15.

Related Body Corporate means related bodies corporate as defined in section 50 of the Corporations Act.

Relevant Authority means:

- (a) any government department or Minister;
- (b) the Council or any other local government;
- (c) any government or statutory authority; or
- (d) any other person under a Law who or which has a right to impose a requirement, or whose consent is required, with respect to the Development Land or the Site Works.

Revised Proposal has the meaning given to it in clause 8.3(h).

Security Interest means a security interest within the meaning of section 12(1) of the *Personal Properties Securities Act* (Cth).

Services means the services outlined in the External Infrastructure Works Design Brief.

Site means the land located at Cowpasture Road, Wetherill Park NSW, being land contained in:

(a) Lot 18, 19, 20, 21 and 22 in DP13961;

(b) part Lot 2 in DP1212087; and

(c) part Lot 17 in DP13961 (Northern Basin land),

as shown in the plan in Schedule 2 having a land area of 226,102m² (excluding the area of the Northern Basin).

Site Construction Licence means the document in the form set out in Annexure C.

Site Works means the External Infrastructure Works and the Internal Site Works combined.

Statutory Costs means all statutory charges and taxes which are payable in relation to the Development Land or the relevant Development Site including:

(a) rates, assessment, or other fees, levies or charges (for example Council rates, water rates, sewerage rates); and

(b) land tax,

but excludes stamp duty.

Take-up Fee means the amounts payable by the Developer to the Owner under clause 11.2 as calculated in accordance with Part 3 of Schedule 4.

Take-up Target means the area set out in the second column of the table contained in Part 3 of Schedule 4 that is to be subject to a Construction Licence or Ground Lease.

Take-up Target Date means the date that coincides with the time after the Effective Date referred to in the first column of the table contained in Part 4 of Schedule 4.

Tender means the Owner's Request for Proposals WSPT2017/34 dated November 2017 in respect of the development and lease of the Site and the **Developer's** (or a Related Body Corporate of the Developer's) response to that request for proposal.

Term means the term commencing on the Effective Date and terminating in accordance with clause 2.2.

Termination Event has the meaning given in clause 20.1.

Undeveloped Land means that part of the Development Land which is not at the relevant time subject to a Construction Licence that has commenced between the Owner and the Developer or a Ground Lease between the Owner and the Approved Tenant.

Updated Development Masterplan has the meaning given to that term in clause 12.1(b).

Upfront Development Payment means

Upfront Land Payment means the upfront land payment for the Development Land or Development Site calculated in accordance with Part 1A of Schedule 4 as adjusted in accordance with this Agreement. **Urban Farming Irrigation Water Reuse Scheme** means the irrigation scheme described in clause 3.7.

Western Sydney Parklands has the meaning given to it in the WSPT Act.

WSPT Act means the Western Sydney Parklands Act 2006 (NSW).

1.2 Construction

Unless expressed to the contrary:

- (a) words importing:
 - (i) the singular include the plural and vice versa; and
 - (ii) any gender includes the other genders;
- (b) if a word or phrase is defined cognate words and phrases have corresponding definitions;
- (c) a reference to:
 - a person includes a firm, unincorporated association, corporation and a government or statutory body or authority;
 - a person includes its legal personal representatives, successors and assigns;
 - a statute, ordinance, code or other law includes regulations and other statutory instruments under it and consolidations, amendments, re-enactments or replacements of any of them;
 - (iv) a Right includes a benefit, remedy, discretion, authority or power;
 - (v) an Obligation includes a warranty or representation and a reference to a failure to observe or perform an Obligation includes a breach of warranty or representation;
 - (vi) provisions or terms of this Agreement or another document, agreement, understanding or arrangement include a reference to both express and implied provisions and terms;
 - (vii) time is to local time in Sydney;
 - (viii) "\$" or "dollars" is a reference to the lawful currency of Australia;
 - (ix) anybody (other than a party to this Agreement) which ceases to exist, is reconstituted, renamed or replaced or has its powers transferred, refers to the body established in its place or which serves substantially the same objects as or succeeds to its powers;
 - this or any other document includes the document as varied or replaced and notwithstanding any change in the identity of the parties; and
 - (xi) writing includes any mode of representing or reproducing words in tangible and permanently visible form, and includes facsimile transmission;

- (d) where any form of the word "include" appears, it is to be read as if followed by the words "without limitation"; and
- (e) a reference to this Agreement includes all schedules and annexures referred to in it.

1.3 Headings

Headings do not affect the interpretation of this Agreement.

1.4 Severability

- (a) If any provision of this Agreement or its application to any person or circumstance is or becomes invalid illegal or unenforceable the provision shall so far as possible be read down to such extent as may be necessary to ensure that it is not illegal invalid or unenforceable.
- (b) If any provision or part of it cannot be so read down the provision or part of it shall be deemed to be void and severable and the remaining provisions of this Agreement shall not in any way be affected or impaired.

2A Conditions subsequent

2A.1 Condition subsequent

- (a) This Agreement is subject to and conditional upon the Owner obtaining the Minister's consent to:
 - (i) this Agreement; and
 - (ii) all of the transactions contemplated by this Agreement, including the Agreements for Ground Lease and Ground Leases,

on terms and conditions acceptable to the Owner and the Developer.

(b) Clauses 2, 3, 4, 5, 6, 7, 8, 9, 10, 11, 12A, 12, 13, 14, 15 and 16, and the Site Construction Licence, will have no effect and the parties will have no obligations under such clauses or the Site Construction Licence until the condition subsequent in clause 2A.1(a) is satisfied.

2A.2 The Owner to use reasonable endeavours

- (a) The Owner must use reasonable endeavours to obtain the Minister's consent to satisfy the condition subsequent under clause 2A.1(a) as soon as reasonably practicable after the date of this Agreement, and on terms and conditions acceptable to the parties.
- (b) The Owner must give written notice to the Developer stating that the Minister has provided his consent within 5 Business Days after it receives such consent from the Minister. The Owner must provide a copy of the Minister's consent together with the notice to the Developer. The Owner must state in its notice whether or not the Minister's consent is acceptable to the Owner.
- (c) Within 4 Business Days of the later of the date the Developer receives:
 - (i) the notice in clause 2A.2(b) confirming the Minister has given his consent; and
 - (ii) a copy of the Minister's consent,

the Developer must give written notice to the Owner as to whether or not the Minister's consent is acceptable to the Developer.

- (d) The parties must accept the Minister's consent under clauses 2A.2(b) and 2A.2(c) if the Minister's consent is substantially in the Form of Ministerial Consent.
- (e) If the Minister's consent is not substantially in the Form of Ministerial Consent, then either party may accept or reject the Minister's consent under this clause 2A.2 at its absolute discretion.
- (f) If either party does not accept the Minister's consent under this clause 2A.2, the Owner must use reasonable endeavours to obtain the Minister's consent on terms and conditions acceptable to the parties. Clauses 2A.2(b) to 2A.2(e) will apply again to any further Minister's consent that the Owner obtains.
- (g) If the Owner does not obtain the Minister's consent on terms and conditions acceptable to the parties in accordance with this clause 2A, by the date that is 60 days from the date of this Agreement, then either the Owner or the Developer may terminate this Agreement by notice to each other party to this Agreement. Any termination of this Agreement under this clause 2A.2(g) will be without prejudice to any preceding breach of this Agreement.

2 Appointment and term

2.1 Appointment

The Owner appoints the Developer as the exclusive developer in accordance with this Agreement for the Development of the Development Land and the Developer accepts that appointment.

2.2 Term

- (a) The Term commences on the Effective Date and expires on the earlier of:
 - (i) ;
 - the date the Owner has entered into Ground Leases for the whole of the Development Land other than areas not capable of being developed; and
 - (iii) any sooner determination of this Agreement.

2.3 Exclusivity

86164821 - 180555 (PXB)

- (a) The Owner acknowledges that the Developer is the exclusive developer for the Development Land and the Owner must not, during the Term:
 - (i) deal with any person other than the Developer in relation to the Development and leasing of the Development Land (or any part of the Development Land) except the Approved Tenant; or
 - develop itself or allow another person to develop the Development Land (or any part of the Development Land) except the Approved Tenant.
- (b) During the Term, the Developer has an exclusive right, subject to the provisions of this Agreement, to:
 - (i) market the Development Land, in conjunction with the Owner;

- (ii) develop Development Proposals;
- (iii) negotiate Agreements for Ground Lease on the same terms as the Pro-Forma Agreement for Ground Lease (subject to this Agreement); and
- (iv) carry out and manage the Development on the Development Land

("Development Rights").

- (c) The Owner must, as soon as practicable, refer any persons who approach the Owner regarding any proposed Development of the Development Land to the Developer.
- (d) This clause 2.3 does not prohibit or restrict the Owner from dealing with any government or statutory authority in relation to any obligations of the Owner under this Agreement or from referring leasing or other queries in relation to the Development Land to the Developer.

3 The Developer's role and responsibilities

3.1 Marketing and management responsibilities

- (a) The Developer must;
 - develop and implement a marketing strategy for the Development Land, which may include advertising if this is appropriate and necessary to properly promote the Development Land;
 - spend a minimum of the constraint (in aggregate) during the Term on marketing activities (including the provision of the Developer's internal architectural and graphic design services) for the Development Land;
 - (iii) provide the Owner with written evidence of the Developer's expenditure on marketing activities;
 - (iv) undertake marketing activities for the Development Land in accordance with the marketing strategy prepared by the Developer under this clause 3.1(e); and
 - (v) pay all marketing and promotional expenses incurred in the course of complying with clause 3.1(a)(i) and clause 3.1(a)(iv).
- (b) As soon as reasonably practicable after the Effective Date, the Developer must prepare and submit details of its marketing strategy for the Development Land to the Owner for approval.
- (c) The Owner:
 - (i) must review the marketing strategy;
 - (ii) may request further information in respect of the marketing strategy or may request reasonable amendments to the marketing strategy;
 - (iii) must notify the Developer of whether it approves or does not approve the marketing strategy within 10 Business Days of the date of submission of the marketing strategy by the Developer,

and if it does not approve the marketing strategy, must give reasons for its decision; and

(iv) if a marketing strategy is not approved by the Owner, the Developer must re-submit to the Owner a revised marketing strategy containing proposals to address the reasons for the Owner's rejection and the provisions of this clause 3.1(c) apply to the revised marketing strategy.

The Owner must not unreasonably withhold its approval of the marketing strategy.

- (d) The Developer agrees to undertake its marketing activities for the Development Land in a manner generally consistent with the marketing strategy approved by the Owner, unless otherwise approved by the Owner (such approval not to be unreasonably withheld).
- (e) If the Developer wants to change the marketing strategy, then the Owner must review the changes to the marketing strategy and clauses 3.1(c) and 3.1(d) will apply.

3.2 Leasing responsibilities

The Developer must:

- (a) actively promote the Development Land to potential tenants during the Term for the Approved Permitted Use;
- (b) use reasonable endeavours to source tenants that meet the Development Proposal Criteria for the Development Land;
- (c) undertake all design and other preparatory work necessary to develop Development Proposals;
- (d) negotiate Agreements for Ground Lease and Ground Leases with Approved Tenants for parts of the Development Land in accordance with this Agreement;
- (e) organise execution of Agreements for Ground Lease and Ground Leases prepared in accordance with this Agreement; and
- (f) (if required) collect the Approved Tenant's security deposits or bank guarantees, and provide such items to the Owner.

3.3 Reimbursement of costs

The Developer must comply with all of its obligations under this Agreement at its own cost and expense in all respects, except to the extent that a right of payment or reimbursement is expressly provided under this Agreement.

3.4 Access Road Costs

In this Agreement,

- (a) **Access Road** means the internal access road within the Site (including associated stormwater drainage).
- (b) Despite any other provision of this Agreement, if during the Term in carrying out a Development in accordance with this Agreement, the Developer causes damage to the Access Road then the reasonable cost of rectification works will be paid by the Developer.

3.5 Drainage Basin Costs

- (a) Subject to clause 3.5(b), the Developer is responsible for the repair and maintenance of the Drainage Basins until the later of:
 - (i) the completion of the Drainage Basin Works (including expiration of the relevant defects, rectification period under clause 6.7) and
 - (ii) the earlier of:
 - (A) commencement of the final Ground Lease; and
 - (B) expiry of the Term.
- (b) To the extent that council or any other Relevant Authority is not responsible for maintaining the Drainage Basins, the Owner will, after the date referred to in clause 3.5(a)(ii)(B) at the cost of the tenants under the Ground Leases (in accordance with the Ground Leases), maintain Drainage Basins included in the Drainage Basin Works. The Owner must use reasonable endeavours to minimise the costs to ensure that the amount of those costs is fair and reasonable having regard to the condition of the Drainage Basins.
- (c) If during the Term:
 - in carrying out a Development, the Developer causes damage to the Drainage Basins (other than by fair wear and tear) after completion of the Drainage Basins; and
 - (ii) as a result of the damage referred to in clause 3.5(c)(i), additional work in respect of the ongoing repair and maintenance of the Drainage Basins is required,

then the additional works will be undertaken by the Owner and the costs of those works must be paid by the Developer to the extent that those costs were incurred as a result of the damage referred to in clause 3.5(c)(i).

- (d) The Owner will provide the Developer with such information and documentary evidence as the Developer may reasonably require to evidence the actual expenditure by the Owner pursuant to clause 3.5(c).
- (e) The Developer must reimburse the Owner for the costs referred to in clause 3.5(c) within 30 days of demand.
- (f) The Owner must use its reasonable endeavours to minimise the costs referred to in clause 3.5(c) to ensure that the amount of those costs is fair and reasonable having regard to the condition of the Drainage Basins prior to the damage caused by the Developer.

3.6 Biodiversity Corridor Mitigation Measures

- (a) In designing and carrying out the Development, the Developer will undertake biodiversity corridor mitigation measures as set out in this clause 3.6:
 - ensure significant green belts are retained and rejuvenated on Lots 5 and 6 so that at least 1.6 hectares are retained as biodiversity corridors connecting through to the Western Sydney Parklands;and

- (ii) incorporate a minimum landscape setback of 22 metres consisting of dense native vegetation in accordance with the landscape plans with a view to soften the visual aspect to the eastern boundary of the Site.
- (b) The Developer agrees to:
 - (i) ensure that no additional remnant vegetation, including Cumberland Plain Woodland (CPW), over and above that detailed in the Initial Development Consent is removed from the Site;and
 - (ii) incorporate a revised basin layout with a more natural form to complement the landscaping and vegetation corridors in line with the Western Sydney Parklands Biodiversity Strategy 2012 -2020.
- (c) The parties agree that any cost-savings captured as a result of the decrease in landscaped area are to be used by the Developer to increase the specification of the remaining adjacent landscaping area in an effort to increase the density and quantity of landscaping and vegetation over and above the levels identified in Planning Documentation.
- (d) The Developer agrees to take reasonable steps to mitigate any no visual impacts on third parties, including residential properties, resulting from amendments to the Planning Documentation.

3.7 Urban Farming Irrigation Water Reuse Scheme

- (a) The Developer has agreed to provide an Urban Farming Irrigation Water Reuse Scheme.
- (b) The concept strategy is to store an additional 3,000m³ of stormwater for the proposed water irrigation scheme on the Owner's land.
- (c) The Developer has agreed to expend a capital sum of \$250,000 to construct a pump station and reticulation, in accordance with the design and plan included in Schedule 5 or an alternative strategy agreed between the Owner and the Developer.
- (d) Subject to clause 3.7(e), the Developer is responsible for the repair and maintenance of the Urban Farming Irrigation Water Reuse Scheme until the later of:
 - (i) the completion of the Urban Farming Irrigation Water Reuse Scheme (including expiration of the relevant defects rectification period under clause 6.7); and
 - (ii) the earlier of:
 - (A) commencement of the final Ground Lease; and
 - (B) expiry of the Term.
- (e) The Owner will, after the date referred to in clause 3.7(d) at the cost of the tenants under the Ground Leases (in accordance with the Ground Leases), maintain the Urban Farming Irrigation Water Reuse Scheme. The Owner must use reasonable endeavours to minimise the costs to ensure that the amount of those costs is fair and reasonable having regard to the condition of the Urban Farming Irrigation Water Reuse Scheme.

(f) The Developer will be responsible, at its expense, for obtaining all necessary irrigation licences and other Approvals required for the construction and operation of the Urban Farming Irrigation Reuse Scheme.

3.8 Developer's warranties

The Developer represents and warrants to the Owner that:

- (a) it will use all due skill and care in carrying out its obligations under this Agreement;
- (b) it has the necessary expertise and resources to carry out and complete the Development or to cause the Development to be carried out and completed in accordance with this Agreement; and
- (c) the Development will be carried out in a proper and workmanlike manner under adequate and competent supervision and in any case in accordance with:
 - (i) all applicable Law;
 - (ii) the relevant Approvals;
 - (iii) this Agreement and relevant Agreement for Ground Lease (if applicable); and
 - (iv) the relevant Australian Standards and the National Construction Code in respect of the Internal Site Works; and
- (d) materials used will be new and of merchantable quality in respect of the Internal Site Works.

4 The Owner's role and responsibilities

4.1 Not used

4.2 The Owner's role and responsibilities

The Owner must:

- (a) consider Development Proposals put forward by the Developer in good faith and in a prompt and reasonable manner, subject to the other provisions of this Agreement;
- (b) enter into Agreements for Ground Lease in accordance with this Agreement;
- (c) enter into Ground Leases in accordance with this Agreement;
- (d) as and when required under this Agreement, enter into a Construction Licence;
- (e) pay or reimburse the Developer for the amounts specified under this Agreement as being payable or reimbursable to the Developer;
- (f) pay all Statutory Costs and Outgoings in relation to the Development Land. However, the Owner may cease to pay Statutory Costs and Outgoings in respect of a Development Site from the commencement date of the Construction Licence, or the commencement date of the Ground Lease, in respect of that Development Site, whichever is the earlier. Despite the previous sentence, the parties agree that the

Developer is not responsible for, and is not required to pay any land tax (including any back dated charges, penalties or fees) charged or payable on the whole or any part of the Development Land in respect of a period prior to the commencement date of the Construction Licence, or the commencement date of the Ground Lease, in respect of each Development Site, whichever is the earlier.

4.3 Removal of expired covenants

- (a) The Owner authorises the Developer to negotiate with the relevant counterparties and do all other things necessary to procure removal of any expired registered covenants and any obsolete easements from the relevant titles of the Development Land ("Obsolete Encumbrances").
- (b) The Owner must:
 - (i) put the relevant certificates of title on deposit at LRS to facilitate the removal of the Obsolete Encumbrances from title; and
 - (ii) provide any assistance reasonably required by the Developer to procure removal of the Obsolete Encumbrances including signing any consents or deeds necessary to remove the Obsolete Encumbrances provided those consents and deeds are in a form acceptable to the Owner (acting reasonably).
- (c) The Developer must pay the Owner's reasonable legal costs and disbursements incurred pursuant to clause 4.3(b).

5 DMA Sunset Date

5.1 Owner's right to terminate

- (a) If one or more Approved Tenants have not entered into Agreements for Ground Lease for a minimum of for the Development Land by the DMA Sunset Date, the Owner may within 60 days after that date (but not otherwise) terminate this Agreement and neither party will have any liability to the other, except in relation to any prior breaches.
- (b) The Developer acknowledges that if this Agreement is terminated under this clause 5.1, the Owner is entitled to retain the Upfront Development Payment and any instalments of the Take-up Fee paid under this Agreement up to the date of termination.

5.2 Extension of DMA Sunset Date

- (a) Subject to clause 5.3, if:
 - (i) the Developer gives to the Owner a written claim for extension of the DMA Sunset Date stating:
 - (A) the fact or likelihood of delay in the Developer procuring one or more Approved Tenants to enter into Agreements for Ground Lease for a minimum of (in aggregate) of the Development Land;
 - (B) the cause of the delay;
 - (C) the date on which the cause of the delay first arose and the date on which the delay ceased; and
 - (D) the number of days of extension claimed; and

(ii) there is a delay and the cause of the delay is an Allowable Delay Event,

then the Developer is entitled to an extension of the DMA Sunset Date, and the Owner must, within 20 Business Days of the Developer's notice under clause 5.2(a), give:

- (iii) written notice to the Developer of any extension of the DMA Sunset Date granted; or
- (iv) if the claimed extension is not granted, written notice of and the reasons for that decision.
- (b) If the Owner:
 - (i) gives a notice under clause 5.2(a)(iv) stating it does not agree with the Developer's notice under clause 5.2(a); or
 - does not give a notice within the relevant time period under clause 5.2(a),

the Owner is deemed to dispute the Developer's notice under clause 5.2(a), and the parties must resolve the dispute in accordance with clause 19.

(c) The DMA Sunset Date cannot be extended beyond the date that the Term expires under clause 2.2.

5.3 Conditions precedent to extension of DMA Sunset Date

- (a) The Developer is only entitled to an extension of the DMA Sunset Date under clause 5.2 if:
 - the Developer gives notice under clause 5.2 within 30 days after it has become aware of the fact or likelihood of delay;
 - (ii) the Developer has used reasonable endeavours to minimise the delay; and
 - (iii) the Developer is continuing to use reasonable endeavours to procure Approved Tenants to enter into Agreements for Ground Lease.
- (b) The extension of the DMA Sunset Date will be either:
 - (i) if the Owner agrees with the Developer's notice under clause 5.2(a), as set out in the Developer's notice under clause 5.2(a); or
 - (ii) if clause 5.2(b) applies, as determined under clause 19.

5.4 Delay caused by the Owner

lf:

- (a) the Developer is granted an extension of the DMA Sunset Date under clause 5.2; and
- (b) the cause of the delay is a result of the event contemplated in paragraph
 (b) of the definition of Allowable Delay Event,

then:

(c) the Developer is entitled to an abatement or reimbursement of the Takeup Fee (if any) payable or paid during the period of that delay calculated on a pro rata basis in the proportion that the area of the Development Land affected by the delay bears to the total Developable Site Area.

6 Site Works

6.1 Principal Contractor

- (a) The Owner agrees that the Developer is the principal contractor for the Site Works and authorises the Developer to:
 - (i) have management or control of those parts of the Development Land on which the Site Works are performed; and
 - discharge the duties imposed on the principal contractor for the Site Works under the Work, Health and Safety Regulations 2011 (NSW).
- (b) The Developer:
 - (i) accepts that it is the principal contractor under clause 6.1(a); and
 - shall discharge the duties imposed on the principal contractor for the Site Works under the Work, Health and Safety Regulations 2011 (NSW) (regardless of whether that engagement is effective).

6.2 External Infrastructure Works

- (a) The Developer must carry out the External Infrastructure Works at its cost.
- (b) The Developer must ensure the design and construction of the External Infrastructure Works:
 - (i) are undertaken by employees, consultants and subcontractors with appropriate professional qualification and experience;
 - (ii) are in accordance with the External Infrastructure Works Design Brief;
 - (iii) are in accordance with all Laws;
 - (iv) are in accordance with all Approvals; and
 - (v) are developed using skilled design consultants with relevant expertise.
- (c) The Developer is responsible for (and will control, co-ordinate, administer and direct) all activities necessary for the completion of the External Infrastructure Works including, without limitation:
 - the appointment, procurement and removal of builders, contractors, tradesmen, development managers, architects, quantity surveyors, engineers and all other consultants, but no such appointment operates to relieve the Developer of its obligations under this Agreement; and

- (ii) the obtaining of all Approvals required to complete the External Infrastructure Works.
- (d) The parties agree that, the Developer will undertake and complete such parts of the External Infrastructure Works as are required to facilitate the construction and use of the first Development Site prior to the practical completion of the first Development Site.
- (e) Nothing in this Agreement obliges the Developer to undertake the External Infrastructure Works until such works are required for the Nominated Developer to undertake and complete a Development.
- (f) For the avoidance of doubt, the Owner agrees that the Developer may undertake the External Infrastructure Works and Internal Site Works at any time prior to the time that such works are required for the Developer to undertake and complete a Development, including before the Developer submits a Development Proposal to the Owner in relation to the relevant Development Site.
- (g) For the avoidance of doubt, the Developer may, but is not obliged to, carr out, provide or install temporary services for construction purposes, gas services or remediation works, in connection with carrying out the External Infrastructure Works.

6.3 Internal Site Works

- (a) Subject to clause 6.3(d), the Developer must carry out the Internal Site Works relevant or required for each Development Site at its cost:
 - (i) progressively and in sequence in accordance with clause 6.3(b); and
 - (ii) in accordance with designs and specifications developed by the Developer, in accordance with the Development Masterplan (including any Updated Development Masterplan) and the other requirements of this Agreement.
- (b) The parties agree that, subject to clause 6.3(d), the Developer will undertake Internal Site Works as such works are required in order to undertake the Development of each Development Site.
- (c) Nothing in this Agreement obliges the Developer to undertake the Internal Site Works until such works are required for the Nominated Developer to undertake and complete a Development.
- (d) For the avoidance of doubt, the Owner agrees that the Developer may undertake the Internal Site Works for each Development Site at any time prior to the time that such works are required for the Nominated Developer to undertake and complete a Development, including before the Developer submits a Development Proposal to the Owner in relation to the relevant Development Site.
- (e) For the avoidance of doubt, the Developer may, but is not obliged to, carry out any other works as are reasonably necessary to enable the Internal Site Works to be undertaken.

6.4 Design and construction of Site Works

(a) The Developer must, either itself or through consultants engaged by it, undertake the design of the Site Works, and the Developer must ensure that the Design Documentation in respect of such Site Works:

- (i) are undertaken by employees, consultants and subcontractors with appropriate professional qualification and experience;
- (ii) have regard to and are generally consistent with an Approved Development Proposal in respect of the relevant Development Site;
- (iii) are generally consistent with the Development Masterplan;
- (iv) are in accordance with all Approvals and all applicable Laws;
- (v) are developed using skilled design consultants with relevant expertise; and
- (vi) are in accordance with the relevant Australian Standards and the National Construction Code.
- (b) The Developer must, either itself or through consultants engaged by it, undertake the construction of any Site Works, and the Developer must ensure the construction of the Site Works:
 - (i) are undertaken by employees, consultants and subcontractors with appropriate professional qualification and experience;
 - (ii) have regard to and are generally consistent with the requirements of the approved Development in respect of the relevant Development Site;
 - (iii) are generally consistent with the Development Masterplan;
 - (iv) are in accordance with all Approvals and all applicable Laws;
 - (v) are in accordance with the Design Documentation in respect of the Internal Site Works;
 - (vi) are in accordance with the relevant Australian Standards and the National Construction Code; and
 - (vii) are undertaken with materials that are new and of merchantable quality.

6.5 Practical completion of works

The parties agree that the superintendent or project manager in respect of the Site Works will certify (as an independent certifier) to the Owner and the Developer that such works have reached Practical Completion.

6.6 Variations to the Site Works

Subject to clauses 6.4(a)(ii), 6.4(a)(iii), 6.4(a)(iv) and 12.1, the Developer may make any changes to the Design Documentation in respect of the Site Works that:

- (a) it considers necessary or desirable; or
- (b) are required by a Relevant Authority.

6.7 Defects rectification

(a) The Developer must repair or rectify, within a reasonable time after notice has been given by the Owner, any defects or faults in the External Infrastructure Works or Internal Site Works due to faulty design or faulty materials or workmanship which are notified in writing to the Developer within 12 months after Practical Completion of the External Infrastructure Works and Internal Site Works as the case may be. This clause does not limit any liability that the Developer may have under this Agreement or at Law in relation to defects or faults in relation to the External Infrastructure Works or Internal Site Works.

(b) The provisions of this clause 6.7 apply in relation to any repair or rectification work carried out by the Developer under clause 6.7(a) for a period of 12 months after practical completion of such repair or rectification work.

6.8 Estimated Internal Site Works costs

As at the Effective Date, the parties agree that the estimate of the Internal Site Works cost for the Internal Site Works is the Internal Site Works Budget.

6.9 Internal Site Works Budget and Program

- (a) The Developer must prepare and submit to the Owner a draft budget in respect of the costs of the Internal Site Works for the Development Land ("Draft Budget") and a draft program showing the proposed time table for carrying out of the Internal Site Works ("Draft Program"):
 - (i) as soon as reasonably practicable after the Effective Date but in any event prior to the first submission of a Development Proposal under clause 8;
 - (ii) at meetings of the Project Control Group; and
 - (iii) not less than once every 6 months during the Term,
- (b) The Draft Budget must set out the following:
 - (i) the Internal Site Works cost incurred to date;
 - the Developer's estimate of future Internal Site Works cost for completion of the Internal Site Works yet to be undertaken (including the Developer's estimate of the Internal Site Works costs for the Internal Site Works for the next Development Site);
 - (iii) the Developable Site Area as at the date of the Draft Budget; and
 - (iv) the Developer's calculation of the amount of the increase or decrease in the total Internal Site Works Budget in comparison to the estimated Internal Site Works costs, on both a total cost basis and rate per square metre basis,

and must include:

- supporting calculation in respect of the Developer's estimate of future Internal Site Works Cost.
- (c) The Draft Program must set out the following:-
 - (i) the Internal Site Works carried out to date for each Development Site; and
 - (ii) the Developer's estimated time table for the carrying out of future Internal Site Works.

- (d) The Owner:
 - (i) must review each Draft Budget and Draft Program submitted to the Owner; and
 - (ii) may request further information from the Developer in respect of the Draft Budget and Draft Program submitted to the Owner, and the Developer must provide such other requested information.
- (e) If the total cost of the Internal Site Works is less than the Internal Site Works Budget, the Developer must pay to the Owner an amount equal to 50% of the difference on the commencement date of the Ground Lease in respect of the last Development Site undertaken by the Developer. The Owner is not responsible for any increase in the cost of the Internal Site Works beyond the Internal Site Works Budget.
- (f) The parties agree that any program under this clause 6.9 is provided for information only and the Developer is not required to carry out the Internal Site Works in accordance with the programs.

6.10 Health Safety and Environmental matters

- (a) In carrying out the Internal Site Works, the Developer will, subject to clause 17, be responsible for complying with all relevant Environmental Laws, and relevant work, health and safety Laws.
- (b) The Developer must prepare and submit to the Owner a draft environmental management plan and a draft work health and safety management plan, prepared in accordance with best industry practice:
 - (i) as soon as reasonably practicable after the Effective Date but in any event prior to the first submission of a Development Proposal under clause 8; and
 - (ii) once every 6 months during the Term.
- (c) The Owner:
 - must review each draft environmental management plan and each draft work health and safety management plan submitted to the Owner;
 - (ii) may (acting reasonably) request further information from the Developer in relation to the draft plans submitted to the Owner, and the Developer must provide such other requested information; and
 - (iii) must notify the Developer of whether it approves (acting reasonably) or does not approve (acting reasonably) the draft plans within 10 Business Days of the date of submission of the draft plans, and if it does not approve a draft plan, must give reasons for its decisions.
- (d) The Owner must not unreasonably withhold its approval of the draft plans.
- (e) If a draft plan referred to in clause 6.10(b) is not approved by the Owner, the Developer must resubmit to the Owner a revised plan containing proposals to address the reasons for the Owner's rejection and the provisions of clause 6.10 apply to the revised plan.

- (f) The Developer must promptly notify the Owner in writing of any environmental management incidents and work health and safety incidents, must investigate such incidents, must report the results of such investigations in writing to the Owner and must implement such action as may be reasonably required to rectify such incidents and to ensure that such incidents are not repeated.
- (g) When requested to do so, the Developer must make available to the Owner or its consultants its records in relation to environmental management and work health and safety management of the External Infrastructure Works and Internal Site Works.

6.11 Site Construction Licence

- (a) The Developer and the Owner must enter into the Site Construction Licence on the date of this Agreement.
- (b) Despite any other provision of this Agreement or the Site Construction Licence, the Developer and the Owner agree that on the commencement date of each Construction Licence for a Development Site, the definition of "Land" under the Site Construction Licence is varied to exclude the relevant Development Site;
- (c) For the avoidance of doubt, the Site Construction Licence is not a "Construction Licence" for the purposes of this Agreement and any Agreements for Ground Lease, Construction Licences and Ground Leases.

7 Development

7.1 Supply of the Development Land for development

- (a) The Owner agrees to supply the Development Land to the Developer in accordance with this Agreement for the Development in accordance with the Development's demand for the Development of the Development Land.
- (b) The Owner agrees to make available the Development Land to the Developer for Development at the Effective Date.

7.2 Staging

- (a) The parties' intention at the Effective Date is that the majority of the Internal Site Works will be undertaken in one stage.
- (b) Notwithstanding clause 7.2(a), the parties agree that retaining walls between lots and final earthworks may be carried out in stages so as to allow variations both in location and levels to suit the specific requirements of each Development.

8 Developments on the Development Land

8.1 Submission of Development Proposals

- (a) The Developer must use reasonable endeavours to prepare and submit to the Owner Development Proposals during the Term.
- (b) The parties agree that:
 - the Developer may, at any time, submit a Development Proposal nominating itself or a Nominee as a proposed tenant in respect of a Development Site to enter into an Agreement for Ground Lease with the Owner; and

- clause 8.1(b)(i) does not in any way limit the Developer's ability to nominate third parties as a proposed tenant in respect of a Development Site.
- (c) The Developer must regularly report to the Project Control Group on its activities in seeking and procuring Development Proposals.
- (d) The Developer and Owner must liaise and co-operate with each other in relation to the development of Development Proposals.
- (e) The Developer and the Owner agree a target of direct employment on average across all Development Sites is for the owner's of Development Land (full and part time). The Developer must, upon the Owner's reasonable request during the Term but not more than once per year, use reasonable endeavours to provide to the Owner information about employment across the Development during the Term, to the extent the Developer has access to that information. For the avoidance of doubt, the Developer is not required to incur additional costs or collect information specifically for the purpose of satisfying its obligations under this clause.

8.2 Development Proposal

A Development Proposal must:

- (a) indicate whether the party to be the "Developer" under the Agreement for Ground Lease and the Construction Licence will be the Developer or a Nominee, and if a Nominee, details of the Nominee;
- (b) include a heads of agreement setting out the commercial terms for an Agreement for Ground Lease and Ground Lease including particulars of the proposed tenant (and guarantor, if any) under the Ground Lease;
- (c) indicate the proposed permitted use and GFA under the Ground Lease;
- (d) include the details of the proposed Lease Value Rate;
- (e) include the Developer's calculation of the proposed Lease Value;
- (f) include the proposed Initial Ground Rent and proposed Upfront Land Payment for the Development Site;
- (g) include the draft Design Documentation for the proposed Development (to the extent available);
- (h) include the dates or times for commencement and completion of construction and occupation of the Development Site;
- (i) include plans and specifications of the Development Site;
- (j) include details of the Internal Site Works and the Building Works required to be completed for the Development;
- (k) include details of any requested variations to the External Infrastructure Works;
- include audited financial statements of the proposed tenant (and guarantor, if any) under the Ground Lease (unless a new entity supported by a parent entity) and any holding company of the proposed tenant for the period 2 years preceding the date of the relevant Development Proposal;

- (m) include a diagram showing the corporate relationship between the proposed tenant (and guarantor, if any) and other companies within the relevant corporate group;
- (n) include the proposed size of the area of the Development Site; and
- (o) include the **Developer's estimate of the cost of the relevant** Development, together with:
 - (i) reasonable details of the Developer's calculation of that estimate; and
 - (ii) the Developer's estimate of the amount of the "Construction Bank Guarantee" under the Agreement for Ground Lease.

8.3 Development Proposal Criteria

- (a) The Owner must notify the Developer whether it approves or does not approve a Development Proposal or a Revised Proposal within 10 Business Days of the date the Developer submits a Development Proposal under clause 8.1(a) or within 10 Business Days of the date the Developer submits a Revised Proposal under clause 8.3(g)(i).
- (b) If:
 - a Development Proposal submitted under clause 8.1(a); or
 - (ii) a Revised Proposal submitted under clause 8.3(h),

satisfies the Development Proposal Criteria, and would not require any material changes to the Pro-Forma Agreement for Ground Lease or Proforma Construction Licence, then the Owner must approve (by giving written notice to the Developer) that Development Proposal or Revised Proposal. The Owner must not take into account the provisions of clause 8.1(e) in considering a Development Proposal or Revised Proposal submitted by the Developer under this clause 8.

- (c) If a Development Proposal or a Revised Proposal does not comply with the Development Proposal Criteria, the Owner must approve or reject that Development Proposal or a Revised Proposal. If the Owner rejects a Development Proposal or Revised Proposal, the Owner must give written details of the grounds for the rejection.
- (d) For the avoidance of any doubt, the Owner's approval of any Development Proposal or a Revised Proposal which does not comply with the Development Proposal Criteria is not to be taken in any way as a waiver of or amendment to the Development Proposal Criteria.
- (e) If a Development Proposal or Revised Proposal satisfies the Development Proposal Criteria, other than the criteria listed in paragraph (f) of Schedule 1, then the Owner must (subject to clause 8.3(f)) act reasonably in deciding to approve or reject the Development Proposal.
- (f) If the Developer submits a Development Proposal or Revised Proposal which includes:
 - (i) a proposed permitted use other than the Approved Permitted Use; or
 - (ii) residential use in any Development Site,

then the Owner has an absolute discretion in approving the proposed permitted use.

- (g) The Developer may include in a Development Proposal:
 - details of additional internal roads and an appropriate adjustment to the area of the Site and Development Land and the Owner must act reasonably in deciding to approve or reject the additional internal roads and area of the Site and Development Land (except that where such changes will have a negative financial impact on the Owner the Owner will be deemed not to have acted unreasonably in rejecting any additional internal roads); and
 - (ii) an updated Development Masterplan and the process in clause 12.1 will apply to the updated Development Masterplan.
- (h) If a Development Proposal is rejected by the Owner, the Developer may re-submit to the Owner for the Owner's consideration a revised Development Proposal containing proposals to address the reason for rejection ("Revised Proposal"). The provisions of this clause 8 apply to a Revised Proposal.
- (i) The Developer must promptly notify the Owner in writing if it becomes aware of any change in circumstances such that the proposed tenant (combined with any proposed guarantor) might no longer satisfy the Development Proposal Criteria in clause 1 of Schedule 1 or might no longer fall within the definition of Approved Tenant in clause 1.1. Notwithstanding anything to the contrary in this Agreement, the Owner is not obliged to enter into an Agreement for Ground Lease with a proposed tenant who (combined with any proposed guarantor) does not satisfy the Development Proposal Criteria in clause 1 of Schedule 1 or no longer falls within the definition of Approved Tenant in clause 1.1.

8.4 Owner to enter into the Agreement for Ground Lease, Ground Lease and Construction Licence

- (a) In respect of an Approved Development Proposal:
 - the Developer must procure the drawing up of the Agreement for Ground Lease and Ground Lease by the Developer's solicitors for the relevant Development Site the subject of the Approved Development Proposal:
 - (A) in the form of the Pro Forma Agreement for Lease as completed or amended as required to:
 - (aa) include the Approved Tenant's guarantor, if any; and
 - (ab) fill in any other blank spaces or do anything else to the Pro Forma Agreement for Ground Lease and Ground Lease necessary to complete the documents;
 - (B) to include the terms of the Approved Development Proposal;
 - (C) the Agreement for Ground Lease is to be between the Owner (as landlord), the Nominated Developer (as developer) and the Approved Tenant;

to include the Approved Tenant's (and guarantor's, if any) limitation of liability clauses (if relevant) provided that the Agreement for Ground Lease and Ground Lease include warranties by the Approved Tenant (and guarantor, if any) in their capacity as trustee, responsible entity or custodian of the Approved Tenant (and guarantor, if any) that it is the holder of the assets of the relevant trust, that (in the case of the Approved Tenant) its interest in the leased property will be an asset of the relevant trust and that the Approved Tenant (and guarantor, if any) is and will be entitled to be indemnified out of the assets of the relevant trust in relation to its obligations under or in relation to the Agreement for Ground Lease and Ground Lease (but subject to the usual exclusions):

(D)

- (E) to include the relevant Upfront Land Payment and Initial Ground Rent payable by the Approved Tenant in the Agreement for Ground Lease and Ground Lease subject to increase as provided in the Agreement for Ground Lease; and
- (F) to include the relevant "Architect" nominated by the Developer at the time of entering into the Agreement for Ground Lease;
- the Developer must procure the drawing up of the Construction Licence by the Developer's solicitors for the relevant Development Site the subject of the Approved Development Proposal:
 - (A) in the form of the Pro Forma Construction Licence; and
 - (B) the Construction Licence is to be between the Owner and the Nominated Developer;
- (iii) the Developer must, in conjunction with the Owner and the Owner's solicitors:
 - (A) finalise the Agreement for Ground Lease, Ground Lease and Construction Licence;
 - (B) have the final version of the Agreement for Ground Lease, Ground Lease and Construction Licence prepared and executed by the Nominated Developer, Approved Tenant and/or any guarantor of the Approved Tenant (as required); and
 - (C) deliver the Agreement for Ground Lease, Ground Lease and Construction Licence, duly executed by the Nominated Developer, the Approved Tenant and any guarantor of the Approved Tenant (as required), and any bank guarantee or other security required under the Agreement for Ground Lease or Ground Lease to the Owner;
- (iv) the Developer must, when submitting an Agreement for Ground Lease, Ground Lease or Construction Licence to the Owner for execution, ensure that marked up versions of those documents are submitted showing all differences between the Pro-Forma Agreement for Ground Lease and Pro-Forma Construction Licence and the proposed final form of those documents;

- (v) subject to clause 8.3(i), the Owner must:
 - (A) finalise the Agreement for Ground Lease, Ground Lease and Construction Licence, promptly, and in good faith;
 - (B) execute the Agreement for Ground Lease, Ground Lease and Construction Licence promptly after receiving the Agreement for Ground Lease, Ground Lease and Construction Licence from the Developer for execution;
 - (C) grant a Ground Lease to an Approved Tenant as and when required under the relevant Agreement for Ground Lease; and
 - (D) comply with the Owner's other obligations under each Agreement for Ground Lease, Ground Lease and Construction Licence (including obtaining necessary mortgagee or chargee consents); and
- (vi) the Owner must consider any amendments to the Agreement for Ground Lease, Ground Lease or Construction Licence requested by the Approved Tenant.
- (b) The Initial Ground Rent will commence to be payable on, and the Ground Lease will commence on, the first to occur of:
 - the date the Approved Tenant commences to use the premises to be leased pursuant to the Ground Lease for the permitted use under the Ground Lease; and
 - (ii) the later of:
 - (A) the first to occur of:
 - (aa) the day after the date of Practical Completion of the Development on the Development Site; and
 - (ab) the alternative date specified in the Approved Development Proposal as the date on which Initial Ground Rent will commence (if applicable); and
 - (B) the day after Practical Completion of the Internal Site Works (or such of those works as are required to be completed to facilitate the construction of the Development and the use of the premises to be leased pursuant to the Ground Lease) is achieved; and
 - (C) the day after Practical Completion of the External Infrastructure Works (or such of those works as are required to be completed to facilitate the construction of the Development and the use of the premises to be leased pursuant to the Ground Lease) is achieved.

8.5 Compliance with Agreement for Ground Lease

- (a) The Developer must:
 - (i) comply with its obligations under each Agreement for Ground Lease and Construction Licence (where applicable);

- (ii) not do or allow anything to be done which would or might cause the Owner to be in breach of any Agreement for Ground Lease, Construction Licence or Ground Lease;
- (iii) notify the Owner promptly if the Developer is in breach of any obligation under an Agreement for Ground Lease or Construction Licence (where applicable) or if the Developer becomes aware of any dispute in relation to an Agreement for Ground Lease or Construction Licence (where applicable) or the performance of the Developer's duties under an Agreement for Ground Lease or Construction Licence (where applicable);
- (iv) notify the Owner promptly if any representation or warranty made or taken to be made by or on behalf of the Developer in connection with an Agreement for Ground Lease or Construction Licence (where applicable) is found to be incorrect or misleading when made or taken to be made; and
- (v) notify the Owner promptly if it becomes aware that any other party to an Agreement for Ground Lease or Construction Licence is in material breach of any term, condition or obligation under that document.

(b) The Owner must:

- (i) comply with its obligations under each Agreement for Ground Lease and Construction Licence;
- (ii) not do or allow anything to be done which would or might cause the Developer to be in breach of any obligation under an Agreement for Ground Lease or Construction Licence; and
- (iii) execute, enter into or grant the Ground Lease as required under the relevant Agreement for Ground Lease.

8.6 Termination of Agreement for Ground Lease

If an Agreement for Ground Lease is subsequently rescinded or terminated prior to the commencement date of the Ground Lease due to any act, default or insolvency of the Approved Tenant, then despite anything else in this Agreement:

- (a) if directed by the Owner, the Developer must immediately cease the relevant Development; and
- (b) the Developer may source and submit a new Development Proposal for the Development Site under clause 8, and if that Development Proposal becomes an Approved Development, then the Developer must carry out and complete the Development contemplated by that Approved Development Proposal.

8.7 Construction Licence

The Construction Licence must:

- (a) be for a term commencing on the date the Developer under the Agreement for Ground Lease pays the Net Land Payment to the Owner and notifies the Owner that it has paid the Net Land Payment to the Owner; and
- (b) include payment of a licence fee equivalent to the Construction Licence Fee by the Developer from the Commencement of the Construction Licence in accordance with clause 8.7(a) above; and

(c) include any necessary changes agreed between the relevant parties.

8.8 Measurement of the Development Site

When the proposed Development Site has been determined the Developer will at its cost engage a surveyor to measure the ground area and advise the Developer and the Owner of the area and dimensions of the relevant Development Site.

8.9 Multi-party Side Deed

- (a) The parties agree that the Developer and an Approved Tenant may (and is entitled to) grant a security interest (including a Security Interest) over its rights under this Agreement, an Agreement for Ground Lease and a Construction Licence to its financier or financiers.
- (b) If requested by the Developer or an Approved Tenant, the Owner must enter into a deed or deeds with the Developer, Approved Tenant and the Approved Tenant's and/or Developer's financier in a form reasonably required by the Approved Tenant's and/or the Developer's financier ("Multi-Party Side Deed").
- (c) The Multi-Party Side Deed referred to in clause 8.9(b) above will generally be consistent with the principles contained in Schedule 8 or must otherwise be on terms reasonably acceptable to the parties.
- (d) The parties agree to act reasonably and in good faith in relation to things requested by the Developer or Approved Tenant or their respective financier or financiers, including the negotiation and execution of a Multi-Party Side Deed. Without limiting the previous sentence, the Owner accepts that the Approved Tenant's and/or the Developer's financier may have requirements that are different to the principles contained in Schedule 8 (including the financier's form of Multi-Party Side Deed) and the Owner must act reasonably and in good faith in relation to the financier's requirements.

8.10 Ownership of fixtures

- (a) The Owner acknowledges that the Nominated Developer will undertake and complete the relevant Development in accordance with the relevant Agreement for Ground Lease and the Approved Tenant will pay certain amounts to the Nominated Developer for the construction and completion of the Development.
- (b) Other than as required by Law, and subject to clause 6.3 of the Pro-Forma Agreement for Ground Lease, the parties acknowledge that all improvements forming part of the Development that are constructed on the relevant Development Site will be owned by the Approved Tenant until expiry or earlier determination of the relevant Ground Lease.

9 Approvals

9.1 The Developer responsible

With the exception of the Initial Development Consent, the Developer must use its reasonable endeavours at its cost to obtain all Approvals required to lawfully:

- (a) enter into and properly execute, perform, observe and discharge its rights and obligations under this Agreement;
- (b) plan, design, execute and complete each Development;
- (c) plan, design, execute and complete the Building Works;

- (d) plan, design, execute and complete the Internal Site Works; and
- (e) plan, design, execute and complete the External Infrastructure Works,

and the Owner, if so requested by the Developer, must co-operate with and assist the Developer in obtaining all Approvals, including promptly executing each application for an Approval upon request by the Developer.

9.2 Approval Process

- (a) In order to remove any doubt, the Developer and the Owner acknowledge and agree that:
 - (i) Subject to clause 9.2(b)(ii), the normal development process for any particular Development Site is that:
 - (A) the Developer will lodge a Development Proposal and obtain the Owner's approval to such proposal on the terms and conditions of this Agreement and for the purposes of obtaining the Approved Development Proposal; and
 - (B) if an Approved Development Proposal is obtained and the relevant parties enter into an Agreement for Ground Lease, the Developer will obtain all other Approvals required by Law, including a development consent from the Relevant Authority in relation to the Site Works for the relevant Development Site ("Site Works Development Consent").
 - (ii) The Developer must provide a copy of the Site Works Development Consent to the Owner promptly after it is issued to the Developer.
 - (iii) The Developer must, within 20 Business Days of receipt of the relevant Site Works Development Consent from the Council, notify the Owner whether or not the relevant Site Works Development Consent is on conditions acceptable to the Developer and the Approved Tenant. The Developer may only reject conditions of the Site Works Development Consent if:
 - (A) the conditions materially adversely affect the Developer's ability to comply with its obligations under this Agreement, an Agreement for Ground Lease or an agreement for lease with a future occupier of the Development Site; or
 - (B) compliance with the conditions of the relevant Site Works Development Consent will result in an increase of more than 5% of the aggregate Building Works and the Internal Site Works costs in the application.
 - (iv) If the Developer gives notice under clause 9.2(a)(iii) that the Site Works Development Consent is on conditions which are not acceptable to the Developer, the Developer may require that steps be taken to object to or appeal against the relevant conditions in which case the Owner and the Developer must cooperate with each other in pursuing such objection or appeal without delay. The Developer must pay the Owner's reasonable expenses in connection with such objection or appeal. If after any objection or appeal process required by the Developer has been exhausted, the Site Works Development Consent is still

subject to conditions which are not acceptable to the Developer, having regard to the provisions of clause 9.2(a)(iii), the Developer may require that the Development of the relevant Development Land not proceed until such time as an Site Works Development Consent is obtained on conditions which are acceptable to the Developer having regard to the provisions of clause 9.2(a)(iii).

- (b) Despite clause 9.2(a), and for the avoidance of doubt, the Owner agrees that the Developer may:
 - (i) obtain the relevant Site Works Development Consent at the same time as the Developer obtains the Approvals (including the development consent) for the relevant Development; and
 - prepare and submit applications for development consent from the Relevant Authority in relation to proposed Site Works for each Development Site before the Developer submits a Development Proposal to the Owner in relation to the relevant Development Site.

9.3 Development of a Development Site

If the Developer obtains all relevant Approvals for the Development on terms acceptable to it, the Developer must procure the design and construction of the buildings and works on the Development Site, including the External Infrastructure Works in accordance with the Approved Development Proposal, the Agreement for Ground Lease and all applicable Laws. The Developer must commence and complete the construction of the works on the Development Site within the time frames specified in the Approved Development Proposal.

10 Construction

10.1 Complete Development

The Developer shall complete each Development or cause each Development to be completed in accordance with:

- (a) this Agreement;
- (b) the Development Masterplan, subject to minor variations from the Development Masterplan or other variations from the Development Masterplan approved by the Owner in accordance with this Agreement;
- (c) the Approved Development Proposal, subject to minor variations which are not inconsistent with the Approved Development Proposal or other variations approved by the Owner (which approval must not be unreasonably withheld provided the variations are consistent with the requirements of this Agreement); and
- (d) all Approvals and applicable Laws.

10.2 Comply with Requirements

The Developer at its expense must:

- (a) comply with all proper and lawful requirements of any Relevant Authority in relation to each Development; and
- (b) comply with and give all notices and obtain all certificates, licences and permits required by any law and pay and indemnify the Owner against any statutory fees or charges legally payable in respect of each Development; and

(c) for the avoidance of doubt, the Developer must pay all long service leave levies required to be paid by Law in respect of each Development.

10.3 Insurance

- (a) From the date of commencement of any Development until the date of expiry of the defects liability period for such Development, the Developer must keep, or must ensure that its contractor carrying out the Development keeps:
 - contracts work insurance in the name of the Developer with the Owner to be listed as an additional insured;
 - (ii) public liability insurance in the name of the Developer as well as any contractor, with the Owner to be listed as an additional insured for liability arising out of the Developer's or contractor's actions of an amount not less than \$50,000,000.00;
 - (iii) motor vehicle third party property insurance in the name of the Developer in respect of the Developer's passenger motor vehicles;
 - (iv) workers compensation insurance for all workers; and
 - professional indemnity insurance in the name of the Developer for its potential liability in relation to civil liability claims, by reason of any act, error or omission of the Developer which results in an actual or alleged breach of professional duty under this Agreement;
- (b) All insurers for the insurances referred to in clauses 10.3(a)(i), (ii) and (v) must be from a reputable insurer.
- (c) Each insurance policy referred to in clauses 10.3(a)(i) and (ii) must include a cross liability clause under which the insurer agrees to waive all rights of subrogation against each party noted on the policy.
- (d) Prior to commencement of any works, the Developer must produce to the Owner a copy of the certificate of currency of each policy and evidence that such policies satisfy the requirements of this clause 10.3.

11 Payment

11.1 Upfront Development Payment

In consideration of the Developer receiving the Development Rights, the Developer shall pay the Upfront Development Payment to the Owner on the Effective Date.

11.2 Take-up Fee

- (a) The Developer and the Owner have agreed that:
 - (i) the Developer will aim to achieve each Take-up Target by the relevant Take-up Target Date; and
 - (ii) if the Developer fails to achieve the Take-up Targets by the Take-up Target Dates, the Developer will pay certain amounts to the Owner in accordance with this clause 11.2.
- (b) Subject to clause 11.2(f), if the Developer fails to reach a Take-up Target by the Take-up Target Date as is set out in Part 3 of Schedule 4, the Developer must pay the Take-up Fee calculated in accordance with Part

3 of Schedule 4 to the Owner quarterly in advance during the Term in accordance with this clause 11.2.

- (c) At least 15 Business Days prior to the beginning of each quarter of the Term, the Developer must notify the Owner of its calculation of:
 - (i) the Developing Site Areas at the relevant time; and
 - (ii) the Take-up Fee payable by the Developer for the upcoming guarter.
- (d) If the Owner does not agree with the Developer's calculation of the Developing Site Areas or Take-up Fee set out in the Developer's notice under clause 11.2(c), then the Owner must notify the Developer within 10 Business Days of the date of the Developer's notice under clause 11.2(c), and the parties must co-operate in good faith to agree the relevant Developing Site Areas and Take-up Fee.
- (e) The parties agree that:
 - subject to clause 11.2(e)(ii), the calculation of the Developing Site Areas for any relevant time is to be based on the calculation of Developing Site Areas as set out in the Development Masterplan; and
 - the Developable Site Area as set out in the Development Masterplan must be adjusted upon execution of each Agreement for Ground Lease.
- (f) Notwithstanding any other provision of this Agreement, the Developer is not required to pay the Take-up Fee during, or in respect of, the period commencing on the Effective Date and ending on the date that is 12 months after the Effective Date.
- (g) It is agreed that should future Take-up Targets be achieved, any Takeup Fee paid to the Owner in respect of the failure to reach a Take-up Target by a prior Take-up Target Date will be reimbursed to the Developer through credits to future Upfront Land Payments as part of the calculation of Net Land Payments.

11.3 Upfront Land Payment

- (a) The Developer must give prompt written notice to the Owner and Approved Tenant after an Agreement for Ground Lease becomes unconditional. The Approved Tenant will be required to pay the Upfront Land Payment to the Nominated Developer on the earlier of 15 Business Days prior to the earlier of:
 - the anticipated date for Commencement of Construction (as notified by the Nominated Developer to the Approved Tenant); and
 - (ii) the date that is three months after the date the Agreement for Ground Lease becomes unconditional,

in accordance with the Agreement for Ground lease.

(b) As soon as reasonably practicable after the Nominated Developer has received the Upfront Land Payment in respect of a Development, the Developer must notify the Owner:

- (i) when the Nominated Developer has received the Upfront Land Payment from the Approved Tenant; and
- (ii) of the Developer's calculation of the amount of that Upfront Land Payment that is to be paid to the Owner, being the Upfront Land Payment plus or minus any variance in accordance with this agreement ("Estimated Net Land Payment").
- (c) Within 5 Business Days of the date of the Developer's notice under clause 11.3(b), the Owner must notify the Developer whether or not the Owner accepts the Estimated Net Land Payment. If the Owner notifies that it does not accept the Estimated Net Land Payment, or the Owner does not give a notice in accordance with this clause 11.3(c), the parties must refer the matter for determination under clause 19.

(d) The "Net Land Payment" is:

- the amount of the Estimated Net Land Payment if the Owner notifies the Developer that it accepts the Estimated Net Land Payment under clause 11.3(c); or
- the amount determined under clause 19 as the Net Land Payment if the Owner notifies the Developer that it does not accept the Estimated Land Payment under clause 11.3(c) or the Owner does not give a notice in accordance with clause 11.3(c).
- (e) The Owner must give a tax invoice to the Developer within 5 Business Days of the date the Net Land Payment is determined in accordance with clause 11.3(d).
- (f) For the purposes of clause 4.1(b) of the Construction Licence, the Owner must notify the Developer or Approved Tenant in writing of receipt of the Net Land Payment within 5 Business Days of the date the Owner receives the Net Land Payment.
- (g) Despite any other provision of this Agreement:
 - Commencement of Construction must not commence until the Developer has paid the relevant Net Land Payment to the Owner as well as any Take-up Fee due under clause 11.2; and
 - (ii) the Developer must pay the Net Land Payment to the Owner on the later of:
 - (A) the earlier of:
 - (aa) the date that the Nominated Developer (or relevant appointed subcontractor) is ready for Commencement of Construction; and
 - (ab) the date being 3 months after the date the Agreement for Ground Lease becomes unconditional; and
 - (B) 10 Business Days after the Owner provides a tax invoice under clause 11.3(e).

11.4 Leasing Commission

- (a) If:
 - the Owner, without the involvement or assistance of an external real estate agent, procures a non-related third party to enter into an Agreement for Ground Lease with the Owner and Developer ("New Transaction");
 - (ii) no commission is payable to any other agent in respect of the New Transaction; and
 - the Development proposed under the New Transaction is substantially different to the Development of the relevant Development Site proposed in the Development Masterplan,

then the Developer must pay to the Owner the Leasing Commission in respect of that Development.

- (b) The Developer must pay the Leasing Commission to the Owner as follows:
 - (i) 50% of the Leasing Commission when the Agreement for Ground Lease becomes unconditional; and
 - (ii) 50% of the Leasing Commission on commencement of the Ground Lease.
- (c) For the avoidance of any doubt, only one Leasing Commission is payable in respect of a Ground Lease.

11.5 Construction Licence Fee

- (a) The Developer or the Approved Tenant, as directed by the Developer, must pay the Construction Licence Fee to the Owner under the Construction Licence from commencement of the Construction Licence.
- (b) The Construction Licence Fee must be paid in accordance with the relevant Construction Licence.

11.6 Tax invoices

Despite any other provisions in this Agreement, the parties agree that:

- (a) a party's ("Paying Party") obligation to make a payment to any other party ("Receiving Party") only arises upon receipt of a relevant tax invoice from the Receiving Party; and
- (b) the Paying Party must make the relevant payment within 10 Business Days from the date it receives the relevant tax invoice from the Receiving Party.

12A NSW Code and Implementation Guidelines

In addition to terms defined in this Agreement, terms used in clauses 12A.1 to 12A.5 have the same meaning as is attributed to them in the Implementation Guidelines (as published by the NSW Treasury July 2013). The NSW Code and Implementation Guidelines are available at www.industrialrelations.nsw.gov.au.

12A.1 Primary Obligation

- (a) The Developer must at all times comply with, and meet any obligations imposed by:
 - the NSW Code, in undertaking the construction of the Access Road ("Access Road Works"), the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme; and
 - (ii) the Implementation Guidelines in undertaking the Building Works and Internal Site Works.
- (b) The Developer must notify the Construction Compliance Unit (CCU) and the Owner of any possible non-compliance with:
 - (i) the NSW Code, in relation to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme; and
 - (ii) the Implementation Guidelines, in relation to the Building Works and Internal Site Works,

and of remedial action taken, within 24 hours of becoming aware of the possible non-compliance.

- (c) Where the Developer engages a subcontractor or consultant (including, without limitation, the Builder) in relation to the Building Works and Internal Site Works after the Effective Date, the Developer must ensure that the contract imposes on the subcontractor or consultant equivalent obligations to those in clauses 12A.1 to 12A.4, including that the subcontractor or consultant must, at all times comply with, and meet any obligations imposed by:
 - (i) the NSW Code, in relation to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme only; and
 - (ii) the Implementation Guidelines, in relation to the Building Works and Internal Site Works.
- (d) The Developer must not appoint or engage another party in relation to the Building Works and Internal Site Works where that appointment or engagement would breach a sanction imposed on the other party in relation to:
 - (i) the NSW Code, in relation to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme only; and
 - (ii) the Implementation Guidelines, in relation to the Building Works and Internal Site Works.
- (e) For the avoidance of doubt, the Developer is not required to comply with the NSW Code in relation to the Building Works and the Internal Site Works (other than the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme) including any provisions of the Implementation Guidelines that refer to the NSW Code.

12A.2 Access and information

- (a) The Developer must maintain adequate records of compliance with:
 - (i) the NSW Code, in relation to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme only; and
 - (ii) the Implementation Guidelines, in relation to the Building Works and Internal Site Works,

by it, its subcontractors, consultants and Related Entities (as applicable).

- (b) The Developer must allow, and take reasonable steps to facilitate, authorised personnel (including personnel of the CCU) to:
 - (i) enter and have access to Development Sites controlled by the Developer;
 - (ii) inspect any work, material, machinery, appliance, article or facility;
 - (iii) access information and documents;
 - (iv) inspect and copy any record relevant to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme;
 - (v) have access to personnel; and
 - (vi) interview any person,

as is necessary for the authorised personnel to monitor and investigate compliance with:

- (vii) the NSW Code, in relation to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme; and
- (viii) the Implementation Guidelines, in relation to the Building Works and Internal Site Works,

by the Developer, its subcontractors, consultants and Related Entities.

(c) The Developer must agree to, and comply with (and procure its Related Entities agree to, and comply with), a request from authorised personnel (including personnel of the CCU) for the production of specified documents, in relation to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme, by a certain date, whether in person, by post or electronic means.

12A.3 Sanctions

- (a) The Developer warrants that at the time of entering into this Agreement, neither it, nor any of its Related Entities, are subject to a sanction in connection with the NSW Code or Implementation Guidelines that would have precluded it from responding to a procurement process for work to which the NSW Code and Implementation Guidelines apply.
- (b) If the Developer does not comply with, or fails to meet any obligation imposed by:

- (i) the NSW Code, in relation to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme only; or
- (ii) the Implementation Guidelines, in relation to the Building Works and Internal Site Works,

a sanction may be imposed against it in connection with the NSW Code or Implementation Guidelines (as applicable).

- (c) Where a sanction is imposed:
 - (i) it is without prejudice to any rights that would otherwise accrue to the parties; and
 - (ii) the State of NSW (through its agencies, Ministers and the CCU) is entitled to:
 - (A) record and disclose details of non-compliance with the NSW Code or Implementation Guidelines and the sanction; and
 - (B) take them into account in the evaluation of future procurement processes and responses that may be submitted by the Developer, or its Related Entities, in respect of work to which the NSW Code and Implementation Guidelines apply.

12A.4 Compliance

- (a) The Developer bears the cost of ensuring its compliance with:
 - (i) the NSW Code, in relation to the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme only; and
 - (ii) the Implementation Guidelines, in relation to the Building Works and Internal Site Works, including in respect of any positive steps it is obliged to take to meet its obligations under the Implementation Guidelines.

The Developer is not entitled to make a claim for reimbursement or an extension of time from the Owner or the State of NSW for such costs.

- (b) Compliance with the NSW Code and Implementation Guidelines in accordance with this clause 12A does not relieve the Developer from responsibility to perform the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme and any other obligation under this Agreement, or from liability for any defect in the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme or from any other legal liability, whether or not arising from its compliance with the NSW Code and Implementation Guidelines (as applicable).
- (c) Where a change in this Agreement, the Access Road Works, the External Infrastructure Works, the Drainage Basin Works, the Urban Farming Irrigation Water Reuse Scheme is proposed, and that change may, or may be likely to, affect compliance with the NSW Code and Implementation Guidelines, as required by this Agreement, the Developer must immediately notify the Owner (or nominee) of the change, or likely change and specify:

- (i) the circumstances of the proposed change;
- the extent to which compliance with the NSW Code and Implementation Guidelines in accordance with this Agreement will be, or is likely to be, affected by the change; and
- (iii) what steps the Developer proposes to take to mitigate any adverse impact of the change (including any amendments it proposes to a Workplace Relations Management Plan or Work Health and Safety Management Plan).

12A.5 General

The parties agree that clauses 12A.1 to 12A.4 is the Owner's (as a New South Wales Government agency) implementation of the NSW Code and Implementation Guidelines as a result of the Tender and this Agreement.

12 Development Masterplan

12.1 Development Masterplan

- (a) The parties acknowledge that the Development Masterplan is attached at Schedule 3.
- (b) The Developer must prepare an updated Development Masterplan every 6 months after the Effective Date during the Term ("Updated Development Masterplan").
- (c) Each Development Masterplan must include a statement of the Developable Site Area as at the relevant date of the Development Masterplan.
- (d) The Owner may provide its reasonable comments and any reasonable amendments to the draft Updated Development Masterplan submitted by the Developer within 14 Business Days of receipt of that draft Updated Development Masterplan from the Developer.
- (e) The Developer must incorporate any of the Owner's comments or amendments that it considers is appropriate (acting reasonably), and must submit a final Updated Development Masterplan to the Owner for approval. Subject to clause 12.1(f)(iv) and (v), the Owner must not unreasonably withhold its approval. When the Owner has approved the final Updated Development Master Plan, the parties must initial the final Updated Development Masterplan for identification purposes.
- (f) With respect to the Development Masterplan, the parties agree that:
 - the intention is to maintain as much flexibility as possible to meet market demand;
 - (ii) the Development Masterplan or Updated Development Masterplan may be changed by the Developer from time to time, provided:
 - (A) the Developer prepares and submits the amended draft Development Masterplan to the Owner; and
 - (B) the parties comply with the provisions of clauses 12.1(d) and 12.1(e) in respect of the amended draft Development Masterplan;

- subject to clause 12.1(f)(iv) and (v), the Owner must not unreasonably withhold its approval to the amended draft Development Masterplan;
- (iv) if the Developer submits an amended draft Development Masterplan and the permitted use for the Development Site is residential, or is other than the Approved Permitted Use, the Owner has an absolute discretion in approving the amended draft Development Masterplan; and
- no part of the Northern Basin can ever form part of the Development Land.
- (g) Without limiting the above and clause 8.3(f), the parties agree that the Developer may change (including by adding to) the "access roads" shown on the Development Masterplan or the Updated Development Masterplan (including but not limited to the layout or location of the access road) as is required for the Development, provided such change does not materially adversely affect the access to or through the Development Land or require the Owner to procure land or easement rights from third parties or cause a detrimental financial outcome for the Owner.

13 Access

13.1 Owner must permit access to Undeveloped Land

- (a) From the Effective Date, the Owner must permit the Developer and authorised representatives, consultants and agents of the Developer to enter the Undeveloped Land at reasonable times and upon reasonable notice to the Owner for the purposes of:
 - (i) undertaking pre-development investigations, including testing and other investigation of the condition of the site;
 - (ii) undertaking a survey or surveys;
 - (iii) showing prospective investors and users over the site; and
 - (iv) any other purpose requested by the Developer (acting reasonably).
- (b) Despite any other provision of this Agreement, the Developer acknowledges that the Developer's employees, contractors, agents, invitees and visitors may not access the Undeveloped Land before receiving the Owner's or the Developer's site induction training (as applicable). The Owner must make the Owner's site induction training available to the Developer's employees, contractors, agents, invitees and visitors on reasonable notice by the Developer and the parties agree that the Owner is not required to provide site induction training more than once. The Owner authorises the Developer to provide subsequent site induction training in place of the Owner to the Developer's employees, contractors, agents, invitees and visitors who did not attend the site induction training provided by the Owner.
- (c) The Owner must comply and cause its representatives and consultants to comply at all times with the reasonable requirements for the time being of the Developer and not to delay a Development.

13.2 Access to adjoining land

From the Effective Date, the Owner must permit the Developer and authorised representatives, consultants and agents of the Developer to enter, access and occupy any land adjoining any of the Development Land which is also owned by the Owner and which is identified for the purposes of this clause 13.2 on the plan in Schedule 2 ("Adjoining Land") for the purpose of:

- (a) accessing any part of the Development Land;
- (b) undertaking the External Infrastructure Works, Internal Site Works, the Urban Farming Irrigation Water Reuse Scheme and any other necessary works on the Development Land and relevant Adjoining Land; and
- (c) constructing a "batter" on the relevant Adjoining Land; and
- (d) repair and maintenance of the Drainage Basins during the relevant defects and maintenance period.

13.3 Battering

- (a) The Owner agrees that:
 - (i) the Developer will "batter" into those parts of the Adjoining Land identified as affected by batter on the plan in Schedule 2 as a part of the Internal Site Works; and
 - (ii) subject to clause 13.3(b), the Developer is not, and will not be, obliged to remove or do any works to alter the "batter" on the relevant Adjoining Land referred to in clause 13.3(a)(i).
- (b) The Developer must maintain the batters referred to in clause 13.3(a)(i) at its cost up to the date of Practical Completion of the Development on the Development Site relevant to the particular batters. After this time, the batters will be maintained by the Owner at the reasonable cost of the Developer or the Tenants under the Ground Leases.
- (c) The Developer must reimburse the Owner the reasonable cost of repairing any damage caused to batters by the Developer or the **Developer's Agents**.

13.4 Conditions

The Developer must:

- (a) keep the Owner informed in relation to the exercise of its rights under this clause 13;
- (b) not carry out any works on the Adjoining Land without the prior approval of the Owner, such approval not to be unreasonably withheld or delayed; and
- (c) comply with the reasonable directions of the Owner from time to time in relation to its access to and use of the Adjoining Land and the Undeveloped Land.

For the purposes of clause 13.4(b), the Owner has approved the "batter" works referred to in clause 13.3.

13.5 Release and Indemnity

86164821 - 180555 (PXB)

(a) The Developer releases the Owner from any claim, action, damage, loss, liability, cost or expense which the Developer incurs or is liable for in

connection with any damage, loss, injury or death to or of any person or property arising from the Developer's access to or use of the Undeveloped Land or the Adjoining Land and any liability for damage to the Developer's property arising from the Developer's access to or use of the Undeveloped Land or the Adjoining Land.

(b) The Developer indemnifies the Owner against any claim, action, damage, loss, liability, cost or expense which the Owner incurs or is liable for in connection with any damage, loss, injury or death caused or contributed to by the Developer or the Developer's authorised representatives, consultants, employees, agents, contractors or invitees, access to or use of the Undeveloped Land or the Adjoining Land, except to the extent such damage, loss, injury or death caused or contributed to by the act or omission of the Owner or the Owner's authorised representative, consultants, employees, agents, contractors or invitees.

13.6 Undeveloped Land

- (a) Up to the Effective Date, the Owner retains control of the Development Land, and the Undeveloped Land remains at the risk of the Owner and the Owner must:
 - maintain the Development Land in substantially the same condition that it was in at the date of this Agreement;
 - (ii) use reasonable endeavours to secure the Development Land; and
 - (iii) comply on time with all Laws and requirements of any Relevant Authority in connection with the Development Land.
- (b) From the Effective Date, the Developer accepts control of all the Development Land, and the Development Land is at the risk of the Developer and the Developer must:
 - subject to Developments, External Infrastructure Works and Internal Site Works, maintain the Development Land in substantially the same condition that it was at the date of this Agreement;
 - seek the Owner's consent to fence the Development Land (such consent not to be unreasonably withheld to the style of fencing proposed by the Developer);
 - (iii) promptly following the consent of the Owner under clause 13.6(b)(ii), fence the Development Land and use reasonable endeavours to secure the Development Land; and
 - (iv) comply on time with all Laws and requirements of any relevant Authority in connection with the Development Land.

14 Environmental monitoring

14.1 Developer accepts Development Site

- (a) The Developer:
 - (i) accepts any Contamination existing in or on the Site as at the Effective Date;

- (ii) releases the Owner from any liability in respect of any Contamination existing in or on the Site as at the Effective Date; and
- (iii) agrees to pay Environmental Earth Sciences for provision of satisfactory reliance on their Supplementary Detailed Site Investigations Report dated 24 October 2017 Version 4.
- (b) Subject to the Owner complying with its obligations under this clause 14, if any activity carried out by or on behalf of the Developer on a Development Site necessitates remediation of any Contamination, then any remediation works to be carried out by the Developer will be:
 - (i) subject to the consent of the Owner (which must not be unreasonably withheld or delayed); and
 - (ii) at the Developer's cost and risk.
- (c) Without limiting clauses 14.1(a) and (b), the Developer is responsible for the remediation and management of any Contamination existing in or on the Development Land to the extent such Contamination:
 - (i) renders the Development Land unsuitable for commercial and industrial use; and
 - (ii) is not caused by the Owner or the Owner's Agents.
- (d) The Developer must as soon as reasonably practicable remediate and manage any Contamination for which the Developer is responsible to the extent that such Contamination renders the land unsuitable for commercial and industrial use.

14.2 Notice of Contamination

During the Term:

- (a) the Owner must promptly notify the Developer if it becomes aware of the existence of Contamination in, on or emanating from the Development Land to the extent the existence of that Contamination is not already known by the Developer; and
- (b) the Developer must promptly notify the Owner if it becomes aware of the existence of Contamination in, on or emanating from the Development Land to the extent the existence of that Contamination is not already known by the Owner.

14.3 Owner gives no warranties

Except as provided in this clause 14, the Owner gives no warranties and makes no representations as to the physical condition of the Site (including latent conditions and Contamination) or other conditions or existing services or structures on the Site. The Developer will accept the Development Land subject to any physical conditions, services or structures, including any changes to such physical conditions, services or structures during the Term.

15 Project Control Group

15.1 Creation of Project Control Group

- (a) Not later than 10 Business Days from the Effective Date, the parties must establish a Project Control Group, which will consist of:
 - (i) at least two (2) representatives of the Developer; and

86164821 - 180555 (PXB)

Horsley Drive Stage 2 - Development Management Agreement

- (ii) at least two (2) representative of the Owner.
- (b) The purpose of the Project Control Group is to provide a forum for consultation between the Owner and the Developer with a view to:
 - (i) coordinating the efficient implementation of each Development and monitoring the progress of each Development; and
 - (ii) coordinating the efficient implementation of the Site Works and monitoring the progress of the Site Works.
- (c) Either party may request any other person to attend a meeting of the Project Control Group.
- (d) The Developer and the Owner agree to cooperate and use all reasonable endeavours to use the mechanisms of the Project Control Group to achieve mutually acceptable outcomes in relation to all matters discussed at the Project Control Group.

15.2 Meetings of Project Control Group

- (a) The Project Control Group must meet bi-monthly, commencing promptly after its establishment and until Practical Completion of the last Development, or as otherwise agreed.
- (b) Meetings will be minuted by the representative of the Developer. The Developer's representative will chair meetings of the Project Control Group.
- (c) If a member of the Project Control Group is unable to be present at a meeting, the member may authorise another person to attend the meeting in their place, or may participate in the meeting by telephone or other electronic means.
- (d) A quorum for the Project Control Group meeting requires at least one representative of the Developer and at least one representative of the Owner.

15.3 Conduct of Project Control Group meetings

- (a) The role of the Project Control Group is to monitor, identify and discuss:
 - relevant opportunities to procure a Development on the Development Land;
 - (ii) market conditions;
 - (iii) competitor movements and any other factors relevant to this Agreement;
 - (iv) failures to comply with this Agreement by either party;
 - (v) progress of the Development;
 - (vi) current claims for extensions of time, including details of dates submitted, dates determined and extensions granted;
 - (vii) any other relevant matters suggested by either party (both acting reasonably in this regard); and
 - (viii) status of matters in dispute.

- (b) The Developer's representative must provide notice of each meeting to the Owner's representative at least 5 Business Days before each meeting. The notice must contain an agenda for the meeting.
- (c) The Developer must maintain up to date minutes of each meeting of the Project Control Group, and provide a copy of such minutes to the Owner.

15.4 Reporting

- (a) The Developer must, at least 5 Business Days before each meeting of the Project Control Group, provide a written report to the Owner setting out reasonable details in relation to the items listed in clause 15.4(b).
- (b) The Developer must prepare a report for each meeting of the Project Control Group, which includes the following areas of reporting:
 - (i) executive summary;
 - (ii) copy of previous minutes;
 - (iii) marketing and leasing status, including the current opportunities, marketing initiatives and budget, competitor analysis and market conditions;
 - (iv) legal issues;
 - (v) Approvals;
 - (vi) Design Documentation;
 - (vii) construction, including details and progress of the External Infrastructure Works, Internal Site Works and other facilities;
 - (viii) financials;
 - (ix) programme;
 - (x) occupation health and safety issues;
 - (xi) occupational health and safety policies and procedures with regards to principal contractors on the Development Land; and
 - (xii) any other general issues relating to this Agreement.

15.5 Costs of Project Control Group

The Owner and the Developer shall bear their own costs in relation to the Project Control Group.

16 Single Sites

16.1 Single Site for each Development Site

- (a) The parties agree it is intended that each Development Site will be contained in a separate folio identifier to identify each Development Site as a single site and a separate Certificate of Title will be issued in respect of that folio identifier to facilitate the registration of each Ground Lease.
- (b) In respect of each Development Site the Developer must prepare proposed plans of subdivision, consolidation or amalgamation (as

applicable) for each Development Site to ensure each Development Site is contained within a separate folio identifier ("Subdivision Plan").

- (c) In respect of each Subdivision Plan:
 - the Developer must lodge the Subdivision Plan for approval to the Relevant Authority, and must use reasonable endeavours to procure the Relevant Authority's approval to the Subdivision Plan;
 - the Developer must lodge the Subdivision Plan for registration with the LRS and must use reasonable endeavours to procure registration of the Subdivision Plan and issue of a Certificate of Title in respect of the relevant Development Site; and
 - (iii) the Owner must execute all documentation and do all things reasonably required by the Developer to assist the Developer to prepare the Subdivision Plan or procure registration of the Subdivision Plan with the LRS.
- (d) The parties agree that a Subdivision Plan may include easements, covenants and such other encumbrances as reasonably required (including as reasonably required by the Developer or as required by any Authority), affecting the Development Land and adjoining land, to facilitate the relevant subdivision and Development, provided that such easements, covenants and other encumbrances have been approved by the Owner, such approval not to be unreasonably withheld. The Owner must execute all documentation and do all things reasonably required by the Developer in respect of any such required easements, covenants and other encumbrances.
- (e) Without limiting clause 16.1(d), the parties agree that easements to benefit Development Sites and burden adjoining land may be required for the Internal Site Works (including batters) and in relation to the Drainage Basins and the Urban Farming Irrigation Water Reuse Scheme.

16.2 Development Site Contributions or Levies

- (a) The Developer and Owner agree that:
 - the Initial Development Consent requires the applicant to enter a Transport Infrastructure Contributions Deed (**TIC Deed**) with the Roads and Maritime Authority which includes a contribution of \$300,000;
 - (ii) the Owner will enter the TIC Deed with RMS and be responsible for payment of the TIC Deed contribution. Any additional contributions levied as a result of any planning application or other action of the Developer will be the responsibility of the Developer.
 - (iii) the Initial Development Consent requires the payment of developer contributions under section 7.11 of the *Environmental Planning and Assessment Act 1979* (NSW) in respect of the Development Land (Developer Contributions);
 - (iv) any Developer Contributions levied in respect of the External Infrastructure Works, Internal Site Works, the Building Works and any subdivision works to the Site will be the responsibility of the Developer; and

(v) any Developer Contributions levied in respect of the buildings or other improvements or proposed use (and not the subdivision) in relation to an Approved Development Proposal or an Approved Tenant or a particular Development Site are the responsibility of the Developer or an Approved Tenant and must be paid by the Developer or an Approved Tenant.

17 No assignment by the Developer

17.1 Consent

- (a) The Developer shall not assign or dispose of any right, interest, duty or obligation under this Agreement to any person, firm or corporation.
- (b) An assignment of the Developer's interest under this Agreement shall be deemed not to be a breach of clause 17.1(a) if:
 - (i) the Developer has given the Owner at least one month's notice in writing of the proposed assignment together with details of the parties and a copy of all proposed documentation and all other relevant information; and
 - (ii) the Developer is not in default under this Agreement or the Developer is in default under this Agreement and such default has been waived by the Owner; and
 - (iii) the Developer establishes to the Owner's satisfaction that any proposed assignee meets each of the Assignment Tests; and
 - (iv) the Developer, assignee, Guarantor and any guarantor of the proposed assignee (if applicable) enter into a deed in the form reasonably required by the Owner and Developer under which (amongst other things):
 - (A) the assignee agrees to perform all of the Developer's express and implied obligations under this Agreement, including the obligation to indemnify the Owner, and (if applicable) the proposed guarantor agrees to guarantee the proposed assignee's obligations in accordance with clause 24 of this Agreement;
 - (B) the Developer releases the Owner from all obligations under this Agreement from the date of assignment except in respect of any claim(s) that have arisen before the date of assignment;
 - the Owner releases the Developer and Guarantor from all obligations under this Agreement from the date of assignment;
 - (D) despite paragraph (C), if there is an assignment of the Developer's rights or obligations under this Agreement to a Related Body Corporate of the Developer and the Developer does not satisfy the condition in clause 17.1(b)(iii), the Developer and Guarantor acknowledge that they are not released from their obligations under this Agreement or such guarantee as a result of such assignment; and
 - (E) the assignee's and its guarantor's (if applicable) limitation of liability clause is included, provided that the limitation of liability includes warranties by the assignee

86164821 - 180555 (PXB)

and guarantor (if applicable) in their capacity as trustee, responsible entity or custodian of the Developer and its guarantor (if applicable) that it is the holder of the assets of the relevant trust, that (in the case of the assignee) its interest in this agreement and in the premises to be leased pursuant to the Ground Lease will be assets of the relevant trust and that the assignee and guarantor (if applicable) is and will be entitled to be indemnified out of the assets of the relevant trust in relation to its obligations under or in relation to this Agreement (but subject to the usual exclusions); and

- (v) the Owner has given its written consent to the proposed assignment, which consent is not to be unreasonably withheld.
- (c) Despite the provisions of clause 17.1(b):
 - the Developer need not satisfy the condition in clause 17.1(b)(iii) in relation to an assignment of the Developer's rights or obligations under this Agreement to a Related Body Corporate of the Developer (including in relation to a solvent reconstruction); and
 - (ii) if the condition in clause 17.1(b)(iii) is not satisfied, such assignment shall not release or diminish the obligations of the Developer or the Guarantor under this Agreement.
- (d) The parties agree that if the Developer:
 - elects to satisfy clause 17.1(b)(iii) in relation to a proposed assignment to a Related Body Corporate of the Developer (including in relation to a solvent reconstruction); and
 - (ii) satisfies clause 17.1(b)(iii) in respect of such assignment,
 - (iii) then clause 17.1(c) does not apply in respect of such assignment.
- (e) The Owner must execute the deed referred to in clause 17.1(b)(iv) in a timely manner.
- (f) The Developer will pay to the Owner all reasonable costs of and incidental to enquiries concerning the proposed dealing or persons concerned in such dealing and of the perusal, negotiation, preparation and stamping of all documents and obtaining the mortgagee's consent to such documents reasonably required by the Owner and all stamp duty and mortgagee consent fees.

18 No sale by Owner

- (a) The Owner must not:
 - (i) sell, assign, or transfer its interest in the Development Land, or any Development Site or Development or this Agreement; or
 - (ii) do anything which is analogous or in substitution of a sale, assignment or transfer referred to in clause 18(a)(i).
- (b) Clause 18(a) will not apply in relation to any transfer to any government or statutory authority, or any transfer effected by any statute, provided that in the case of a transfer to any government or statutory authority

which is not effected by statute, the Owner procures the proposed new registered proprietor to enter into a deed with the Developer and the Guarantor and all other parties to any Agreement for Ground Lease, Ground Lease or Construction Licence assuming the Owner's obligations under this Agreement and any Agreement for Ground Lease, Ground Lease and Construction Licence and covenanting to comply with the Owner's obligations under this Agreement and any Agreement for Ground Lease, Ground Lease and Construction Licence from the date of transfer.

(c) Any such deed will include covenants by the Developer and the Guarantor to perform the Developer's and Guarantor's obligations under this Agreement and any Agreement for Ground Lease, Ground Lease and Construction Licence from the date of transfer, and a release of the Owner's obligations under this Agreement and any Agreement for Ground Lease, Ground Lease and Construction Licence and the Developer and the Guarantor must and must procure that all other parties to any Agreement for Ground Lease, Ground Lease or Construction Licence, enter into any such deed when requested to do so.

(d) The Owner must not:

- create or allow to exist a mortgage, security agreement, bill of sale, charge, lien or pledge or Security Interest over any or all of the Development Land, Development Site, Development or this Agreement; or
- (ii) do anything which is analogous or in substitution of the items described in clause 18(a)(i).
- (e) The deed referred to in clause 18(c) must also include the Developer's or Guarantor's (or any assignee's or any guarantor of the assignee's (if applicable)) limitation of liability clause, provided that the limitation of liability includes warranties by the Developer, Guarantor, assignee or the guarantor of the assignee (if applicable) in their capacity as trustee, responsible entity or custodian of the Developer, Guarantor (or any assignee or any guarantor of the assignee) (if applicable) that it is the holder of the assets of the relevant trust, that its interest in this Agreement will be assets of the relevant trust and that the Developer, Guarantor, assignee or the guarantor of the assignee is and will be entitled to be indemnified out of the assets of the relevant trust in relation to its obligations under or in relation to this Agreement (but subject to the usual exclusions).

19 Dispute resolution

19.1 Notification

If a dispute between the Owner and the Developer arises out of or in connection with this Agreement, each party must (except in any proceedings for equitable relief, in which case this clause 19.1 does not apply) furnish in writing to the other **party detailed particulars of that party's claim, or**, where the other party is not the claimant, the reasons for rejecting the claimant's claim.

19.2 The reply

The party against whom the claim is made must reply within 14 days after receipt of the detailed particulars.

19.3 The dispute

If the claim is not accepted within 14 days of the party making the claim then a dispute is deemed to exist.

19.4 **Compulsory CEO conference**

Upon a dispute being deemed to exist under clause 19.3, the parties must cause their respective chief executives to meet within 14 days to attempt to resolve the dispute acting reasonably, and if they cannot resolve the dispute to attempt to agree to a process to resolve the dispute between the parties before appointing an expert under clause 19.5.

19.5 Expert

- (a) If the parties fail to resolve the dispute in accordance with clause 19.4, the parties must appoint a person ("Expert") to resolve the dispute. If the parties cannot agree on the appointment of the Expert within 7 days, then the Expert must be appointed by the President of the Australian Resolution Institute, New South Wales Chapter. Either party may approach the President of the Australian Resolution Institute, New South Wales Chapter to appoint the Expert.
- (b) The Expert shall be appointed under such form of expert determination agreement as may be customarily recommended or used by the Australian Resolution Institute for that purpose, or if there is no such form of agreement, then in such form as may be reasonable required by the Expert.
- (c) Upon the Expert being appointed under clause 19.5(a) and accepting the appointment, the parties must direct the Expert to make a determination in relation to the dispute within 14 days of the appointment.
- (d) The parties may make submission to the Expert.
- (e) The Expert must act and make a decision acting as an expert and not as an arbitrator.
- (f) The parties agree that the decision of the Expert will be final and binding upon them.
- The costs of the Expert must be borne equally by the Owner and the (g) Developer.
- (h) The parties must sign all documents and do all things reasonably necessary to effect the appointment of the Expert and to give effect to the intention of this clause 19.

20 Default

20.1 **Termination Events**

- (a) A Termination Event occurs if:
 - the Developer becomes Insolvent; or (i)
 - (ii) the Developer fails to pay any amount to the Owner under this Agreement by the due date for payment; and
- (b) the Owner gives written notice to the Developer in respect of such event and demanding that the event be rectified; and

- (c) the Developer fails to rectify the event within 30 days after the giving of the notice referred to in paragraph (b); and
- (d) the Owner gives further written notice to the Developer demanding that the Developer rectify such event and stating its intention to terminate the Agreement; and
- (e) the Developer fails to rectify the event within 60 days after the giving of the Owners notice under paragraph (d).

20.2 Termination

- (a) If a Termination Event occurs, the Owner may terminate this Agreement by written notice to each other party to this Agreement, without prejudice to any other rights that the Owner may have against any other party to this Agreement.
- (b) For the avoidance of any doubt, the parties agree that a termination of this Agreement under this clause 20 does not and will not affect any Agreement for Ground Lease, Ground Lease or Construction Licence entered into by the parties as at the date of the termination under clause 20.2(a).

20.3 Obligations on termination

- (a) If this Agreement is terminated under clause 20.1:
 - (i) the Developer must remove the Developer's property from the Undeveloped Land; and
 - (ii) the Developer must return the Undeveloped Land to the condition it was in at the date the Developer was given access to it if required by the Owner; and
 - (iii) the Developer must vacate the Undeveloped Land; and
 - (iv) the Developer must give the Owner the keys, access cards and similar devices in respect of the Undeveloped Land held by the Developer, the Developer's employees and agents and any person who has been given them by the Developer or the Developer's employees and agents.
- (b) The Owner may treat the Developer's property on the Undeveloped Land as abandoned and deal with it in any way it sees fit at the Developer's expense if the Developer does not remove its property in accordance with this clause 20.3 or a notice given under it.
- (c) Despite any other provision of this Agreement, the Developer must not remove property which:
 - (i) the Owner has stated (as a condition of giving approval to the works) may not be removed; or
 - (ii) is part of structural work done by the Developer to the Development Land unless the Owner give the Developer a notice requiring the Developer to remove that property.

20.4 No other right to terminate

Despite any Law to the contrary, no party may terminate this Agreement other than in accordance with an express provision of this Agreement.

21 Notices

21.1 General

Any notice, demand, certification or other communication in this Agreement:

- (a) shall be given in writing and in the English language; and
- (b) may be given by an authorised representative of the sender.

21.2 Method of service

In addition to any means authorised by law any communication may be given by:

- (a) being personally served on a party;
- (b) being left at the party's current address for service;
- (c) being sent to the party's current address for service by pre-paid ordinary mail or if the address is outside Australia, by pre-paid airmail;
- (d) being sent by facsimile to the party's current numbers for service; or
- (e) being sent by email to the party's current email address.

21.3 Address for service

(a) The addresses and numbers for service are initially:

THE OWNER

address:	Level 7, 10 Valentine Avenue, Parramatta NSW 2150
facsimile:	+61 2 9895 7580
email:	Kerry.Jahangir@wspt.nsw.gov.au
attention:	Kerry Jahangir
DEVELOPER	
address:	Suite 59 Jones Bay Wharf (Upper Deck) 26-32 Pirrama Road Pyrmont Sydney NSW 2009
facsimile:	+61 2 9506 1499
email:	cipsecretary@ciproperty.com.au
attention:	Company Secretary
GUARANTOR	
address:	Level 20, 1 Martin Place, Sydney NSW 2000
facsimile:	+61 2 8295 8659
email:	chcosecandlegal@charterhall.com.au
attention:	Company Secretary for Charter Hall Holdings Pty Limited

AND

address:	Suite 59 Jones Bay Wharf (Upper Deck) 26-32 Pirrama Road Pyrmont Sydney NSW 2009
facsimile:	+61 2 9506 1499
email:	cipsecretary@ciproperty.com.au
attention:	Company Secretary for Commercial & Industrial Property Pty Ltd

(b) A party may from time to time change its address or numbers for service by notice to the other party.

21.4 Service by post

A communication given by post shall be deemed received:

- (a) if posted within Australia to an Australian address, on the third Business Day after posting; and
- (b) in any other case, on the tenth Business Day after posting.

21.5 Service by facsimile

A communication sent by facsimile shall be deemed received when the sender's facsimile machine produces a transmission report stating that the facsimile was sent to the addressee's facsimile number.

21.6 Service by email

A communication sent by email shall be deemed received when the sender's email system confirms that it has been successfully sent and provided an undeliverable message is not subsequently received by the sender.

21.7 Form received

A communication sent by facsimile shall be deemed given in the form transmitted unless the message is not fully received in a legible form and the addressee immediately notifies the sender of that fact.

21.8 Service after hours

If a communication to a party is received by it:

- (a) after 5.00 pm; or
- (b) on a day which is not a Business Day,

it will be deemed to have been received on the next Business Day.

22A GST

- (c) An amount payable by a party under this Agreement in respect of a taxable supply by the other party, unless expressed to represent the price of supply, represents the value of the supply and the recipient of the supply must, in addition to that amount and at the same time, pay to the supplier the GST in respect of the supply.
- (d) If this Agreement requires a party to pay, reimburse or contribute to an amount paid or payable by the other party in respect of a creditable

acquisition from a third party, the amount for payment, reimbursement or contribution will be the value of the acquisition by the other party plus, if the payment, reimbursement or contribution is consideration for a taxable supply, the GST payable in respect of that supply.

- (e) A party is not obliged under clauses 22A(a) or 22A(b) to pay the GST on a taxable supply to it under this Agreement until that party is given a valid tax invoice for the supply.
- (f) Expressions used in this clause 22A have the meanings given to them in the GST Act.

22 Representations and warranties

22.1 Owner representation and warranties

- (a) The Owner represents and warrants that:
 - (i) it has been duly constituted under the WSPT Act;
 - (ii) it has power to enter into and observe its obligations under this Agreement and all documents and transactions contemplated by this Agreement, including without limitation the Agreement for Ground Lease, Ground Lease and Construction Licence (each a "Transaction Document" and together the "Transaction Documents"); and
 - (iii) it has in full force and effect the authorisations necessary under the WSPT Act to enter into each Transaction Document, observe obligations under each Transaction Document, and allow each Transaction Document to be enforced; and
 - (iv) its obligations under this Agreement and each Transaction Document are valid and binding and are enforceable against it in accordance with its terms; and
 - (v) this Agreement and the Transaction Documents the transactions contemplated under them do not contravene its constituent documents, WSPT Act or any other Law, regulation or official directive or any of its obligations or undertakings by which it or any of its assets are bound or cause a limitation on its powers; and
 - (vi) except as provided in the WSPT Act, it does not hold its interest under this Agreement as trustee.
- (b) The Owner represents and warrants to the Developer that, the Owner has obtained all necessary consents to enter into this Agreement, (including all consents required under the WSPT Act, any finance, security or other documents).

22.2 Developer and Guarantor representations and warranties

The Developer and the Guarantor each represents and warrants that:

- (a) it has power to enter into and observe its obligations under this Agreement and all documents and transactions contemplated by this Agreement; and
- (b) it has in full force and effect the authorisations necessary to enter into this Agreement and all documents and transactions contemplated by this

Agreement, observe obligations under them, and allow them to be enforced; and

- (c) its obligations under this Agreement and all documents and transactions contemplated by this Agreement are valid and binding and are enforceable against it in accordance with its terms; and
- (d) this Agreement, the Agreement for Ground Lease, Ground Lease and Construction Licence and the transactions contemplated under them do not contravene its constituent documents or any law, regulation or official directive or any of its obligations or undertakings by which it or any of its assets are bound or cause a limitation on its powers or, the powers of its directors, to be exceeded.

23 Miscellaneous

23.1 Legal costs

- (a) The Developer agrees to reimburse to the Owner, the Owner's reasonable legal costs in relation to:
 - the negotiation, preparation and execution of any heads of agreement prepared prior to execution of this Agreement; and
 - (ii) the negotiation, preparation and execution of this Agreement (including all schedules and annexures to this Agreement),

up to a maximum amount of \$100,000.00 (in aggregate).

- (b) The Developer agrees to reimburse to the Owner, the Owner's reasonable legal costs in relation to the negotiation, preparation and execution of:
 - each Agreement for Ground Lease up to a maximum amount of \$5,000 per Agreement for Ground Lease;
 - (ii) each Ground Lease up to a maximum amount of \$5,000 per Ground Lease; and
 - (iii) each Construction Licence up to a maximum amount of \$1,500 per Construction Licence.
- (c) Despite clause 23.1(a), the Developer is not obliged to pay any amount to the Owner under this clause until the Owner provides to the Developer a tax invoice in respect of that payment.
- (d) The Developer must pay any amount payable under this clause 23.1 within 14 Business Days of the date the Developer receives a tax invoice from the Owner.

23.2 Stamp Duty

The Developer shall, as between the parties, be liable for and duly pay all stamp duty (including any fine or penalty except where it arises from default by the other party) on or relating to this Agreement and any document executed under it including the Agreements for Ground Lease, Construction Licences and Ground Leases.

23.3 Amendment

This Agreement may only be varied or replaced by a document duly executed by the parties.

23.4 Waiver and exercise of rights

- (a) A single or partial exercise or waiver of a right relating to this Agreement will not prevent any other exercise of that right or the exercise of any other right.
- (b) A party will not be liable for any loss, cost or expense of any other party caused or contributed to by the waiver, exercise, attempted exercise, failure to exercise or delay in the exercise of a right.

23.5 Rights cumulative

Subject to any express provision in this Agreement to the contrary, the rights of a party under this Agreement are cumulative and are in addition to any other rights of that party.

23.6 Further assurance

Each party shall promptly execute all documents and do all things that any other party from time to time reasonably requires of it to effect, perfect or complete the provisions of this Agreement and any transaction contemplated by it.

23.7 Computation of time

Where time is to be reckoned by reference to a day or event, that day or the day of that event is excluded.

23.8 Governing law and jurisdiction

- (a) This Agreement is governed by and is to be construed in accordance with the laws in force in New South Wales.
- (b) Each party irrevocably and unconditionally submits to the non-exclusive jurisdiction of the courts of New South Wales and any courts which have jurisdiction to hear appeals from any of those courts and waives any right to object to any proceedings being brought in those courts.

23.9 Joint and several liability

An obligation of two or more persons binds them jointly and severally.

23.10 Counterparts

This Agreement may consist of a number of counterparts and if so the counterparts taken together constitute one and the same instrument.

23.11 Effect of execution

This Agreement is not binding on any party unless it or a counterpart has been duly executed by, or on behalf of, each person named as a party to the Agreement.

23.12 Entire understanding

- (a) This Agreement embodies the entire understanding and agreement between the parties as to the subject matter of this Agreement.
- (b) All previous negotiations, understandings, representations, warranties, memoranda or commitments in relation to, or in any way affecting, the subject matter of this Agreement are merged in and superseded by this Agreement and shall be of no force or effect whatever and no party shall be liable to any other party in respect of those matters.
- (c) No oral explanation or information provided by any party to another shall:

- (i) affect the meaning or interpretation of this Agreement; or
- (ii) constitute any collateral agreement, warranty or understanding between any of the parties.

23.13 No agency

The parties acknowledge that the relationship of the parties under this Agreement does not constitute one of agency, partnership, trust or joint venture.

23.14 Owner must execute

The Owner must execute all documentation and do all things:

- (a) reasonably required by the Developer in respect of the Development; and
- (b) reasonably required for compliance with any agreement, Law or Approval in respect of the Development,

including without limitation, executing any consents, forms, reports, application or plans.

23.15 Exercise of power

Notwithstanding any other provision of this agreement, the Owner must exercise its power under the WSPT Act and the relevant regulations insofar as they relate to this Agreement or the Developer's rights under this Agreement independently and objectively and without regard to the Owner's interests and rights under this Agreement.

24 Guarantee

24.1 Guarantee

In consideration of the Guarantor entering into this Agreement at the request of the Owner and the Developer, the Guarantor unconditionally and irrevocably guarantees to the Owner the due performance, observance and completion of:

- (a) all the terms, conditions and obligations imposed on the Developer under this Agreement including the due and punctual payment of all monies now or hereafter becoming payable under this Agreement at any time; and
- (b) all of the terms conditions and obligations of:
 - (i) the Developer under this Agreement and the Site Construction Licence;
 - (ii) the Nominated Developer under the Agreement for Ground Lease and Construction Licence;
 - (iii) any assignee of the Developer's obligations under clause 17.1(c) of this Agreement; or
 - (iv) any assignee of the Nominated Developer's obligations under an Agreement for Ground Lease or a Construction Licence to which clause 5.1(c) of the Agreement for Ground Lease applies,

("Developer Entities"), including the due and punctual payment of all moneys now or hereafter becoming payable under this Agreement, each Agreement for Ground Lease and each Construction Licence by the Developer Entities, but only in relation to terms, conditions, obligations and payments to be observed, performed or made in the period up to the Practical Completion of the relevant Development under the relevant Agreement for Ground Lease.

(the "Guaranteed Obligations").

24.2 Indemnity

As a separate and distinct covenant, the Guarantor unconditionally and irrevocably indemnifies the Owner and shall keep the Owner indemnified and held harmless from and against all losses, damages, costs, claims, charges and expenses whatsoever suffered or incurred (whether direct or consequential) by the Owner by virtue of or as a consequence of the breach, default or failure by any of the Developer Entities for any reason whatsoever to duly and punctually perform, observe and complete the Guaranteed Obligations as contemplated in clause 24.1.

24.3 Payment

The Guarantor must pay to the Owner immediately upon demand any amounts due to the Owner under clauses 24.1 and 24.2.

24.4 Performance

If any of the Developer Entities for any reasons fail to duly perform, observe or complete the Guaranteed Obligations, without affecting the generality of the foregoing, the Owner may require the Guarantor to perform, observe or complete this Agreement and each Agreement for Ground Lease and each Construction Licence under the same terms and conditions and the Guarantor shall be responsible for the satisfaction of all those terms and conditions.

24.5 Effect

This guarantee continues in full force and effect until the whole of the Guaranteed Obligations have been satisfied in full unless the Owner in the meantime gives notice in writing to the Guarantor that this guarantee is no longer required.

24.6 Primary obligations

The Guarantor's obligations with respect to the Guaranteed Obligations are primary obligations and the Owner is not obligated to proceed against or enforce any right against any of the Developer Entities or any other party before making a demand on the Guarantor pursuant to this guarantee or otherwise enforcing this guarantee.

24.7 No limitation

Neither the existence of any legal disability on the part of the Developer Entities or the Guarantor or any one or more of them, nor the granting of any time concession or indulgence to, nor the making of any composition of, nor the waiver of any breach or default on the part of any of the Developer Entities, nor the neglect or forbearance of the Owner to enforce this Agreement or any Agreement for Ground Lease or Construction Licence, nor any arrangement or release made between the Owner and any of the Developer Entities or any Approved Tenant with or without the consent of the Guarantor, nor any alteration to or variation of this Agreement or any Agreement for Ground Lease or Construction Licence or the powers rights or obligations of any of the parties, nor assignment of any of the Developer Entities' rights or obligations under this Agreement or any Agreement for Ground Lease or Construction Licence nor any of the Developer Entities' becoming insolvent, defunct or otherwise ceasing to exist, nor any assignment being made of the property of any of the Developer Entities for the benefit of creditors, nor this guarantee being executed in counterparts shall limit, vitiate, void, render voidable, release or discharge this

guarantee or the Guarantor or any one or more of them nor the guarantee in any Agreement for Ground Lease or Construction Licence not being executed by the Guarantor.

24.8 Guarantor liability continues

No sum or sums of money received by the Owner which it may be obliged to refund to any liquidator of any of the Developer Entities or which it may in its discretion refund to any liquidator of any of the Developer Entities shall be considered as received by the Owner or discharge or diminish in any way the liability of the Guarantor under this guarantee.

24.9 Purpose

The Guarantor is a party to this Agreement solely for the purpose of this clause 24.

24.10 Survival

The obligations of the Guarantor under this clause 24 survive the expiry or earlier termination of this Agreement.

24.11 Guarantor may not transfer

The Guarantor may not transfer or otherwise dispose of its obligations under this clause 24.

25 Confidentiality

25.1 Confidential

The Owner and the Developer agree that the contents of this Agreement are sensitive commercial information and that they must use reasonable endeavours to keep the terms of this Agreement confidential.

25.2 Exceptions

The Owner and the Developer will not be in breach of their obligations under clause 25.1 if they give information to:

- (a) a professional adviser;
- (b) an agent or employee;
- (c) a prospective purchaser of the whole or part of the Site;
- (d) a financier;
- (e) an actual or prospective shareholder; or
- (f) any other person as required by law,
- so long as:
- (g) the disclosure is reasonably necessary; and
- (h) the person who discloses the information uses reasonable endeavours to ensure that the person who receives the information keeps the information confidential.

25.3 Overriding Obligations and Rights

(a) Notwithstanding the other provisions of this clause 25, the parties acknowledge and agree that:

- under the GIPA Act, the Owner has obligations to:
 - (A) publicly disclose government information that is open access information (as that term is defined in the GIPA Act), including by publishing this Agreement on the Owner's contract register; and
 - (B) release information pursuant to an access application unless the Owner is reasonably satisfied that there is an overriding public interest against disclosure; and
- (ii) nothing in this clause 25 will limit or otherwise affect the discharge of the Owner's obligations to disclose information in connection with the Development as required by any Law including but not limited to the GIPA Act; and
- (iii) the Developer may request the redaction of commercial in confidence provisions of this Agreement or any other agreement to which the Developer is a party.

26 No merger

(i)

The provisions of this Agreement shall not merge on the commencement or execution of any Agreement for Ground Lease but shall continue in full force and effect after such execution or commencement.

27 Interest

A party must pay the other party interest on any money owing to the first party under this Agreement which is not paid on the due date. The interest payable:

- is at a rate 2% per annum higher than the National Australia Bank Limited ABN 12 004 044 937 Business Lending Indicator Base Rate on the date on which default occurs;
- (b) applies from the day after the money should have been paid to the day that the money is actually paid; and
- (c) is capitalised on the last day of each month.

EXECUTED as an agreement

Signing page

DATED: SO JULY 2018

Owner

Executed on and behalf of the body named below by the authorised delegate(s) whose signature(s) appear(s) below pursuant to the authority specified.

Body: Western Sydney Parklands Trust (ABN 85 202 544 800)

Authority: s 8 of the Western Sydney Parklands Act 2006

Almalate

Signature of witness

ACTER BOARAKATE

Name of witness (print)

Sourcion

Occupation of witness (print)

Developer

EXECUTED by CIP/CH (HORSLEY DRIVE) PTY LIMITED (ACN 6127 \$714 97[) in accordance with section 127(1) of the *Corporations Act* 2001 (Cth) by authority of its directors:

Hacke

Signature of director

RICHARD STACKER

Name of director (block letters)

Signature of delegate

.....

Suellen Fitzgerald Name of delegate

)

)

Position: Executive Director

Signature of director/company secretary* *delete whichever is not applicable

I. A Lynch Name of director/company secretary* (block letters) *delete whichever is not applicable Guarantor

EXECUTED by CHARTER HALL HOLDINGS PTY LIMITED (ABN 15 051 363 547) in accordance with section 127(1) of the Corporations Act 2001 (Cth) by authority of its directors: e of director Name of director (block letters) EXECUTED by COMMERCIAL & INDUSTRIAL PROPERTY PTY LTD (ABN 88 105 081 865) in accordance with section 127(1) of the Corporations Act 2001 (Cth) by authority of its directors: ature of director

Name of director (block letters)

R

Signature of director/company secretary* *delete whichever is not applicable Charisse Nortje

Name of director/company secretary* (block letters) *delete whichever is not applicable

Signature of director/company secretary* *delete whichever is not applicable

Name of director/company secretary* (block letters) *delete whichever is not applicable

Schedule 1 – Development Proposal Criteria

- (a) A proposed tenant under the Development Proposal (combined with any proposed guarantor):
 - (i) in the reasonable opinion of the Owner, having regard to the financial information of the proposed tenant and any proposed guarantor provided by the Developer, is capable of paying (and will have sufficient funds to pay for (whether or not it has debt financing in place)) the cost of proposed development works, the Upfront Land Payment and the Ground Rent; and
 - (ii) has not been convicted of a criminal offence which carries a maximum sentence of equal to or more than 12 months in prison or multiple sentences that add up to equal to or more than 12 months in prison; and
 - (iii) has not been the subject of a formal adverse finding by the Independent Commission against Corruption for conduct which, if prosecuted, carries a maximum sentence of equal to or more than 12 months in prison.
- (b)
- (i) Commencement Date: on the first to occur of:
 - (A) the date the tenant commences to use the premises for the permitted use under the relevant proposed Ground Lease; and
 - (B) the later of:

Ground Lease terms and conditions:

- (aa) the day after Practical Completion of the Development under the relevant proposed Agreement for Ground Lease; and
- (ab) the day after Practical Completion of the Internal Site Works (or such of those works as are required to be completed to facilitate the construction of the Development and the use of the premises to be leased pursuant to the Ground Lease) is achieved; and
- (ac) the day after Practical Completion of the External Infrastructure Works (or such of those works as are required to be completed to facilitate the construction of the Development and the use of the premises under the relevant proposed Ground Lease).
- (ii) Term: 90 years;
- (iii) Rent: equivalent to the Ground Rent calculated for that Development Site;
- (iv) Rent payable: monthly in advance; and

- Outgoings: tenant responsible for outgoings as provided for in the Pro-Forma Ground Lease.
- (c) The proposed Upfront Land Payment has been calculated in accordance with Part 1A of Schedule 4.
- (d) The proposed Upfront Land Payment for the Development Site is for the relevant Lease Value for that Development Site calculated in accordance with Part 1 of Schedule 4.
- (e) The proposed Initial Ground Rent for the Development Site is the Initial Ground Rent for that Development Site calculated in accordance with Part 2 of Schedule 4.
- (f) The proposed permitted use under the Ground Lease is consistent with the Approved Permitted Use.
- (g) Approved Tenant to pay any land tax from the date of Commencement of Ground Lease.
- (h) The date for compliance with conditions precedent for commencement of construction must be not later than 12 months after the date of the Agreement for Ground Lease (as extended in accordance with the Agreement for Ground Lease) and the Owner and the Developer to have the right to terminate the Agreement for Ground Lease if such conditions precedent have not been satisfied within 12 months after the date of the Agreement for Ground Lease (as extended in accordance with the Agreement for Ground Lease).
- (i) The Agreement for Ground Lease is to provide for construction in accordance with relevant Approvals with the following conditions precedent to commencement of relevant works by the Developer or the Approved Tenant:
 - (i) all necessary Approvals for Commencement of Construction obtained, and copies provided to Developer and Owner;
 - (ii) development consent obtained subject to conditions acceptable to Developer, Approved Tenant and Owner;
 - (iii) payment by the Approved Tenant to the Developer of the Upfront Land Payment, and payment by the Developer to the Owner of the Net Land Payment in accordance with this Agreement; and
 - (iv) provision to the Owner of evidence acceptable to the Owner that the Approved Tenant will have funds sufficient to pay for relevant development works and the Ground Rent.
- (j) The Agreement for Ground Lease to include Plans and Specifications that are generally consistent with the Approved Development Proposal.
- (k) The buildings proposed to be constructed on the Development Site must not occupy less than 35% of the land area of the Development Site unless otherwise agreed by the Owner.
- (I) The Development Site must not be less than in size.
- (m) The Development must achieve a minimum 5 star green star rating for the Development.

(n)

The height of any building must not be more than 20 metres, except with the consent of the Owner. The Owner must consider any proposal with a building height of more than 20 metres, having regard to the visual impact on nearby residential areas. The Owner is to assess and acknowledge market trends and technologically driven design changes that may be the driver for heights greater than 20 metres, but its consent may be given or withheld in its absolute discretion. Schedule 2 – Plan of Development Land, Site and Adjoining Land



Schedule 3 – Development Masterplan



Schedule 4 – Calculations

1 Lease Value Rate

1.1 Initial Land Value Rate

The parties agree that the initial land value rate as at the Effective Date for the Development Land is

("Initial Land Value Rate").

1.2 Lease Value Rate

The parties agree that the Initial Land Value Rate is subject to a fixed per annum percentage increase on each anniversary of the Effective Date. The Lease Value Rate is the Initial Land Value Rate increased by per annum under this clause 1.2.

1.3 Lease Value

(a) The parties agree that the Lease Value for a Development Site is calculated as follows:



1A Upfront Land Payment

The parties agree that the upfront land payment will be calculated in accordance with the following formula:



2 Initial Ground Rent

Horsley Drive Stage 2 - Development Management Agreement 30 July 2018



The above example is subject to rounding each figure to the nearest dollar.



3 Take-up Target

(a) The Take-up Target is set out in the table in this paragraph (a) and the Take-up Fee is calculated in accordance with the formula in paragraph (b):



(b) Subject to paragraphs (c) and (d), the Take-up Fee per annum is calculated in accordance with the following formula:



Horsley Drive Stage 2 - Development Management Agreement 30 July 2018



4 Adjustment to Net Land Payment

(a) The parties agree that the Net Land Payment calculated in accordance with clause 11.3 will be adjusted for Take-up Fee reimbursements in accordance with clause 11.2(g).

5 Construction Licence Fee

(a) The Construction Licence Fee payable by the Developer to the Owner in respect of a Development Site is to be paid monthly in arrears and calculated as follows:



For the avoidance of doubt, the monthly Construction Licence Fee is the variable CLF in the above formula divided by 12.



(b)

86164821 - 180555 (PXB)

Horsley Drive Stage 2 - Development Management Agreement 30 July 2018